DEVELOPMENT AND APPLICATION OF CARBON-BASED NANOMATERIALS TO IMPROVE THE PERFORMANCE OF LEAD ACID ENERGY STORAGE DEVICE

by

SREEDHAR NAIDU DORASWAMY

submitted in accordance with the requirements for the degree of

DOCTOR OF PHILOSOPHY

in the subject

MECHANICAL ENGINEERING

at the

UNIVERSITY OF SOUTH AFRICA

SUPERVISOR: Prof. VASUDEVA RAO VEEREDHI

(30th January 2022)

DECLARATION

Name:	Sreedhar Naidu Doraswamy	
Student number:	57650543	
Degree:	Doctor of Philosophy	

Exact wording of the title of the thesis as appearing on the electronic copy submitted for examination:

DEVELOPMENT AND APPLICATION OF CARBON-BASED NANOMATERIALS TO IMPROVE THE PERFORMANCE OF LEAD ACID ENERGY STORAGE DEVICE

I declare that the above thesis is my own work and that all the sources that I have used or quoted have been indicated and acknowledged by means of complete references.

I further declare that I submitted the thesis to originality checking software and that it falls within the accepted requirements for originality.

I further declare that I have not previously submitted this work, or part of it, for examination at Unisa for another qualification or at any other higher education institution.

(The thesis will not be examined unless this statement has been submitted.)

SIGNATURE

___30th January 2022_____ DATE

Abstract

Lead-acid batteries with conventional materials are now no longer able to meet the emerging requirements for several applications in telecom, automobile, solar, etc., Due to the increase in the demand for higher performances, batteries keeping with the change in the CO_2 emissions, carbon-based materials, which are of possible consideration, are investigated to meet the requirements. The areas in which the lead acid battery is more affect are is the sulphate growth or the grid growth of the current collectors and sulphation in the negative electrode.

Several types of carbon with low and higher dimensions have been attempted to study the electrical performances in the negative plates of lead-acid batteries (LAB). Attempts were made to improve discharge capacity, charge acceptance & life cycling ability of the LAB. Out of many types (2D) Graphene has been selected to evaluate the performance of the LAB due to its unique characteristics. The Carbon Black was commercially purchased as the negative active material additive for the performance comparison, and graphene has been synthesized in our laboratory.

Production of Graphene in Bulk form through chemical methods is a significant drawback of graphene-related research, which is hindering its commercialization. In this work, various chemical methods for the synthesis of graphene were examined, like the Hummers method, Modified Hummers method, and Microwave-assisted acid method. The results were very accurate by the microwave-assisted acid method. The productivity and purity of the produced graphene using this method are very high. The Morphology, Structure, and Phase characteristics were well studied by using various characterization techniques like XRD, SEM/EDS, Raman, and BET.

Along with Graphene layers, which were synthesized by microwave-assisted method, the commercially purchased carbon black was also used as part of the electrode material validation for energy applications. The carbon-based nanomaterial is used as an additive for negative electrode material for absorbed Glass Mat (AGM) lead-acid batteries.

For making an electrode for lead acid battery application, electrochemical & electrical studies are essential to rationalize.

In one aspect, to address the premature failure (PMF) due to the grid surface, the graphene material is prepared in the colloidal for and applied to the grids. A 2V cells was assembled to study the coated grid performance and found the positive coated grid has better electrical behavior than the negative coated grids. This higher-rated capacity is prepared with positive coated grids (2V-600Ah). The electrical tests at lower rates 0.1C, 0.5C, and 1C are marginally higher than the uncoated grids. But at higher current discharges for shorter durations like C15 minutes and C5 minutes, the positive coated grids outperformed by 2 to 3 times, respectively.

In another aspect, Graphene was prepared and applied in negative active material of AGM lead acid battery, and studies were carried out. Other individual components of the composite like BaSO₄ & Lignin are used to integrate the low dimensional carbon as an additive composite and are dry mixed with the active material (Lead Oxide) and applied to the lead grid used as current collectors.

By the experimental investigation as per Japanese Industrial Standards (JIS) the electrical performances, it is observed that the current discharge capacity increased by nearly 27% with graphene as an additive in NAM.

Whereas in charge acceptance capability enhancement of 14% is observed, and the time taken for discharge capacity of the batteries with Graphene has exceeded the requirement and for a long-time-interval of 23%, respectively.

The cycle life of the graphene batteries increased by 33% compared to that of the control batteries above the 40%SoH condition.

The results demonstrated that the component particles are uniformly distributed towards the nanomaterials; this tells the proper formation of the composite. Electrochemical studies of graphene nanomaterial electrode reveal that the impedance characteristic is very different from that of traditional one. Therefore, the electrical performances are evidence of these materials having the ability of electrode material properties for energy applications.

Keywords: Carbon; Nanomaterial; Graphene; Lead Acid Battery; Lead Grids; Negative Active Material; Premature Failure (PMF); Progressive Failure (PGF); Sulphation; Stationary Lead Acid Battery; Automotive Lead Acid Battery; Japanese Industrial Standard (JIS); High Current discharges; Energy density (Capacity); Charge Acceptance; Cranking Ability; Charge/Discharge Cycles.

ACKNOWLEDGEMENTS

I sincerely thank my guide Professor Dr. V. Vasudeva Rao, University of South Africa. It was he who motivated me to take up the Doctoral program and introduced me the life changing experience of "Energy". Professor had made me to prepare and drove me through the unknow terrain which later I realised the essence of contributing to the society. He was not just only my guide from a technical perspective but also from a moral perspective. He was always with me even at tougher times and encouraged me to give my best. This has led to several new developments in the field of lead acid batteries.

I am also grateful to Dr RW Maladzhi and Prof. Fulufhelo Nemavhola, Mechanical Engineering, and the Management of University of South Africa for giving me an opportunity to do Doctorate in the Department of Mechanical Engineering and for extending all the necessary support during the course.

My heartfelt thanks to Dr. K S N Murthy, Senior General Manager, Amararaja Batteries Limited, Tirupathi for the unconditional support that he has extended and also for providing necessary inputs at critical junctures during the tenure of the project. Right from commissioning of the test rigs for the development of advanced lead acid batteries for various applications. I can say that by Prof. Vasudeva Rao Veeredhi and Dr. K S N Murthy guidance it was possible to take the research from laboratory scale to industrial. Allowing me to use the facilities and go through the material characterization from building a prototype evaluating it and then taking forward to product validations only happened because of the support from Dr. KSN Murthy. The journey was not just allowing me to use the premises and facilities to establish many test procedures in line to international standards but also target the critical and dynamic market requirements. I wish to extend my gratitude to Mr. M Jagadish, Chief Technology Officer, Amararaja Batteries, and his staff for offering all the necessary support and approvals during the course of the project. I sincerely thank the students at the Sreenidhi Institute of Technology - Hyderabad, Ms. Sandhya, Mr. Deepak, Mr. Narasimha and Mr. Shradul, who were involved in the experimentation work. Their interest, enthusiasm, and most importantly their commitment, even during probing weather conditions, and holidays made me be on my toes all the time.

I would like to thank my Co-Scholars, Mr. Rajeshwar Sripada and Ms. Sandhya, for their support. I wish to thank my fellow colleagues Mr. Lokesh, Mr. Madhava Rao, Mr. Shyamsunder Reddy, Mr. Balakrishnama Naidu, and Mr. Bhoopal who supported me at various stages from electrode preparation, analysis, and device assembly. I would also thank Mr. Rudramurthy, Mr. Sudharsanam, Mr. Prasad. Mr. Hemanth and Mr. Gajendra for their support at electrical testing of the batteries. I also wish to extend my gratitude to Mr. Surendranath reddy, Mr. Srihari, Mr. Hari Prasad and Mr. Jayachand of product development at Amararaja Batteries for their inputs at improving the battery performances inline to the customer requirements. Also, Mr. Senthil and his team from process engineering for their valuable inputs at the time of developing the batteries from cell (2V) to battery scale (12V). I still remember the enthusiasm that they showed during the early days when we had to fight against all the odds to take the concept forward.

During the course of the project work and during the preparation of this manuscript, I referred to several valuable books, journals, and technical papers listed in the References. I wish to thank all the respective authors.

Lastly, and most importantly, I thank my family members for their invaluable contribution for the success of this project. I would like to start with my children Aarna and son Arya Vardhan, who never complained even though they did not get enough attention from me when it was really required. Their smile, achievements, and mischief not only helped me to rejuvenate myself, but also gave me enough strength to withstand the tough times. Thank you could be a small word to use for the contributions made by my wife Shilpa Shree for the successful completion of my thesis. She helped me in all aspects, right from taking care of our children and aging parents, to encouraging me during my down times. She ensured that I don't lose focus during my high times, and most importantly, she helped me manage both office work and research work effectively. My parents never had an opportunity to do higher education but they always wanted me to scale new heights. I am not sure if I have the right word to thank them for the sacrifices they made. In spite of their health conditions, they took the responsibility of taking care of things at home. They never allowed troubles to reach me. My heartfelt thanks to my extended family members viz., my sisters and brothers-in-law for their support to me and to my family members.

Finally, it's my humble duty to thank God, without whose kind support, all this couldn't have been.

NOMENCLATURE

- A Area
- c capacitance
- C15min Capacity Ampere-Hour Rating at 15 minutes
- C5min Capacity Ampere-Hour Rating at 5 minutes
- C1 Capacity Ampere-Hour Rating at 1hour
- C5 Capacity Ampere-Hour Rating at 5 hours
- C8 Capacity Ampere-Hour Rating at 8 hours
- C10 Capacity Ampere-Hour Rating at 10 hours
- CAT Charge Acceptance Test
- CCA Cold Cranking Ampere
- CV Cyclic Voltammetry
- DoD Depth of Discharge
- EV ElectriC Vehicle
- Eº Standard Cell Potential
- GO Graphene Oxide
- H₂O Water
- HEV Hybrid Electric Vehicle
- HLE High Load Endurance
- HRD High-Rate Discharge
- HRPSoC High-Rate Partial State of Charge
- H₂SO₄ Sulfuric Acid
- ICE Internal Combustion Engine
- LAB Lead Acid Battery
- LAESD Lead-Acid Energy Storage Device
- LLE Light Load Endurance
- m = Mass

- NaOH Sodium Hydroxide
- PAM Positive Active Material
- NAM Negative Active Material
- PSoC Partial State of Charge
- Pb Lead
- PbO Lead Monoxide
- PbO₂ Lead Dioxide
- PbSO₄ Lead Sulfate
- Pb(OH)₂ Lead Hydroxide
- Pb₂O₃ Lead Trioxide
- Pb₃O₄ Red Lead
- Q charge
- rGO Reduced Graphene Oxide
- Sb Antimony
- SE Secondary Electron
- SEM Scanning Electron Microscopy
- SLI Starting Lighting Ignition
- SoC State of Charge
- SSA Specific Surface Area (m²/gm)
- Sn Tin
- XRD X-ray Powder Diffraction
- 3BS Tribasic Lead Sulfate
- 4BS Tetrabasic Lead Sulfate
- R Resistance, ohms
- T Temperature, ºC
- v Scan rate
- V Potential Window

LIST OF PUBLICATIONS/CONTRIBUTIONS FROM Ph.D. WORK

Articles:

- 1. Sreedhar Doraswamy, Narendran Dama, Murthy K S N, Jagadish M and Veeredhi Vasudeva Rao "EIS and Electrical Investigations on (1D) Multiwall Carbon Nanotubes as NAM additive for Automotive Lead-Acid Battery" –Communicated to the Journal.
- Sreedhar Doraswamy, K. Srinivas, K.S.N. Murthy, M. Jagadish, Veeredhi Vasudeva Rao, "Effect of milled carbon as negative electrode additive for lead acid energy storage device", Materials Today: Proceedings, Volume 38, Part 5, 2021, Pages 3131-3135, ISSN 2214-7853, https://doi.org/10.1016/j.matpr.2020.09.501. (ELSEVIER).
- Sreedhar D, Vinod Reddy Y, Vasudevarao V "Synthesis and Analysis of Ternary Oxide Nanomaterial for Electro-Thermal Applications" Procedia Materials Science, ELSEVIER Ltd., Vol 10C, Pages 116-123, 2015. (doi:10.1016/j.mspro.2015.06.033)
- Sreedhar D, Srinivasulu Reddy K, Vasudevarao V "Synthesis and Evaluation of Zinc Based Nanomaterials for Optical-thermal Studies" Materials Today Proceedings, ELSEVIER Ltd., Vol. 2, Issues 4–5, 2015, Pages 3690–3696. (doi:10.1016/j.matpr.2015.07.131)
- 5. Shardul B Gaikwad, Sreedhar D, Vasudeva Rao V "Supercapacitor nanocomposite using rGO, MWNT and PANI" IJMITE, ISSN(P) 2348-0513; ISSN(E) 2454-471X, Vol.3, Issue 11, Nov 2015, pp. 1-8.
- Riya Mandal, Sreedhar D, Vasudevarao V "Fabrication of Nano-Hybrid for Constructional Applications" IJMITE, ISSN(P) 2348-0513; ISSN(E) 2454-471X, Vol.3, Issue 11, Nov 2015, pp. 9-14.
- Shardul B Gaikwad, Sreedhar D, K. Srinivasulu Reddy, Vasudeva Rao V "Fabrication and Analysis of Carbon-Based Nanomaterials as Supercapacitor Electrode" (ISBN: <u>978-93-84743-62-8</u>), Bonfring Publications April 2015, AMMA- NIT, Trichy.
- 8. Sreedhar D, Vinod Reddy Y, Srinivasulu Reddy K, Vasudevarao V "Synthesis of Zinc Tin Oxide Nanomaterial at Low Temperature and Study of Its Properties" IJNA, ISSN(P) 2277-4777; ISSN(E) 2278-9391, Vol.4, Issue 6, Dec 2014, Pages. 1-6.
- Sreedhar D, Sandhya Devireddy and Vasudevarao Veeredhi "Facile Synthesis and thermal analysis of Antimony Telluride Nanostructures" Materials Today Proceeding; ELSEVIER Ltd., Vol 5, Issue 2, Part 1, Pages 3403-3410, 2018. (doi: 10.1016/j.matpr.2017.12.089)
- 10.Sreedhar D, Sandhya Devireddy and Vasudevarao Veeredhi "Synthesis and Study of Reduced Graphene Oxide layers under Microwave Irradiation" Materials Today Proceeding; ELSEVIER Ltd., Vol 5, Issue 2, Part 1, Pages 5097-5102, 2018. (doi: 10.1016/j.matpr.2017.11.585)

- 11. Narasimharao K, Venkata Ramana G, Sreedhar D and Vasudevarao V "Synthesis of Graphene Oxide by Modified Hummers Method and Hydrothermal Synthesis of Graphene-NiO Nano Composite for Supercapacitor Application". (doi: 10.4172/2169-0022.1000284)
- 12.K. Srinivasulu Reddy, D. Sreedhar "Thermal Conductivity of Natural Fiber, Glass fiber & CNT's Reinforced Epoxy Composites" International Journal of Current Engineering and Technology (IJCET), Inpressco Ltd., Volume 6, No. 4, 2016. (doi:10.14741/Ijcet/22774106/6.4.2016.18)
- 13.K. Srinivasulu Reddy, D. Sreedhar, K. Deepak Kumar, G. Praveen Kumar "Role of Reduced Graphene Oxide on Mechanical-thermal Properties of Aluminum Metal Matrix Nano Composites" Materials Today Proceedings, ELSEVIER Ltd., Volume 2, Issues 4–5, 2015, Pages 1270–1275. (doi:10.1016/j.matpr.2015.07.042)
- 14. Srinivasulu Reddy K, Sreedhar D, Nagesh P "Fabrication and Mechanical Properties of AlSi12 Nanoparticulate Composite" Procedia Materials Science, ELSEVIER Ltd., Vol 10C, Pages 149-158, 2015 (doi:10.1016/j.mspro.2015.06.036)
- 15. Srikanth A and Sreedhar D "Evaluation of Synthesis of Silver Nanoparticles by Chemical Technology" International Journal of Nanotechnology and Application (IJNA) ISSN 0973-631X, Volume 8, Number 1 (2014), pp. 25-29.

CONTENTS

CHAPTER – 1 INTRODUCTION

1.1	General Background	01
	1.1.1 Lead-Acid energy storage Construction	04
	1.1.2 Electrochemical Reactions of LAB at different stages	08
	1.1.3 Current challenges for Lead Acid Battery	09
	1.1.4 Nanotechnology – A possible solution for Energy Storages	11
1.2	Objective of the study	15
1.3	Scope of the Present Work	16
1.4	Organization of Thesis	17

CHAPTER – 2 LITERATURE REVIEW

2.1	Forms of Carbon	18
2.2	Synthesis of Carbon Nanomaterials – For Energy Storages	20
2.3	Application of Carbon in Lead Acid Battery for progressive failure	21
2.4	Application of Carbon in Lead Acid Battery for premature failure	28
2.5	Gaps Identified	31
2.6	Summary	32

CHAPTER – 3 Problem Definition

3.1	Statement of the problem	35
3.2	Reasons effecting Lead Acid battery performance	35
3.3	Our Approach for possible solutions	38
3.4	Experimental Approach for Premature Failure (PMF)	39
3.5	Experimental Approach for Progressive Failure (PGF)	39
3.6	Summary	40

CHAPTER – 4 Nanomaterial Synthesis

4.1	Synthesis of Carbon Nanomaterials- Existing Methods	41
	4.1.1 Mechanical Exfoliation (Scotch Tape)	41
	4.1.2 Chemical Vapor Deposition (CVD)	42
	4.1.3 Hummers method	43
	4.1.4 Liquid phase exfoliation	44
4.2	Our Approach: Microwave Irradiation	45
4.3	Summary	47

CHAPTER – 5 Material Characterization

5.1	Scanning electron microscopy (SEM)	48
5.2	Transmission Electron Microscope (TEM)	50
5.3	X-Ray diffraction (XRD)	52
5.4	Raman spectroscopy	53

5.5	Resistance Measurement	56
5.6	Surface Area Measurement	58
5.7	Summary	59

CHAPTER – 6 Experimental Process/Investigations

6.1	Manufacturing Process: Lead Acid Battery	60
	6.1.1 Electrode Preparation	61
	6.1.2 Assembly	65
	6.1.3 Acid Filling	67
	6.1.4 Formation & Finishing	68
6.2	Process Validation	70
6.3	PMF: Optical Micrograph	71
6.4	PMF: Energy Dispersive X-Ray Analysis (EDAX)	72
6.5	PGF: Cyclic Voltammetry	74
6.6	PGF: Impedance Spectroscopy	75
6.7	Summary	77

CHAPTER – 7 Results and Discussions

7.1	Application of Graphene to improve Lead Acid Battery Performance	78
	7.1.1 PMF: Electrical Performance of positive coated grids	79
	7.1.2 PGF: Electrical Performance of Graphene in NAM	80
7.2	Summary	86

CHAPTER – 8 Conclusions

8.1	Summary of the present work	89
8.2	Scope for future work	93
Refer	References	
Application and Requirements of LAB: Annexure – A		105
List of Equipment: Annexure - B		108
Other	Other outcomes through this project: Annexure - C	

<u>CHAPTER - 1</u> <u>INTRODUCTION</u>

1.1 General Background

Almost every automobile which runs on IC engines/Electric Vehicles has a Lead-Acid Battery (LAB) as a power source for Starting the engine/motor, Lighting/ Turn on the Electrical equipment's and Drive. In general, a battery is an electrochemical device which store energy in the form of chemical state later converting into electrical form in discharge condition and back to chemical energy at recharge condition. Compared to the available different types of battery technology in society today, Lead-Acid battery is one of the commonly used devices due to its mature technology for reliable energy supply. Usually, Lead-Acid batteries are the most cost-effective and safe in operating technology suitable to various applications like Stationary, traction, Hybrid/Electric vehicles, etc.,

Although the Arsacid empire used electrochemical cells in late 247BC mentioned by A, J. Salkind, the usage of lead/Lead dioxide/Sulfuric Acid was primarily identified with the invention of lead-acid energy storages in 1859 by the renowned French physicist Gaston Plante. However, the foundations for the electrochemical cell discoveries were laid by Luigi Galvani, a medical professional from the University of Bologna, Italy, in the 1780s. Later, Alessandra Volta, in 1796's who had displayed the first primary (Irreversible) cell/battery. To their works and contribution towards electrochemical innovations, the Cell/Battery are termed as "Galvanic / Voltaic Cells." Right from its inception to date, the lead-acid battery is the best-known rechargeable energy storage solution of choice for most automobiles & other industries due to its cost and safety benefits. Lead-Acid Batteries (LAB) are used in a wide range of applications like a power source in a motor vehicle, UPS systems, Telecommunication, and many more applications with increasing demand a constant and uninterruptible source of energy. Usually, Lead-Acid batteries for automotive applications consist of 6 cells, each cell delivering 2 volts, with 12 Volts when connected in series. The cells consist of lead alloy plates that are positive and negative electrodes. Positive and negative plates stacked alternately negative/positive/negative with suitable quality separators placed between the plates to prevent short circuits as essential components shown in the figure below fig.1. The more plates in a cell, the higher the current capacity. The design of Lead-Acid Batteries is simple; the electrodes in each cell are arranged alternatively with one positive and one negative electrode. Each cell group of 6cells electrodes are then connected in a series like Positive to Negative / Negative to positive manner. Electrolyte solution, i.e., dilute sulfuric acid, is added, then chemical action starts. A reaction between the electrodes and the electrolyte creates an electrical charge. The transfer of ions (H⁺ & SO⁻₄) occurs during the battery's charge/discharge. The electrolyte's specific gravity tends to change during the charge/discharge period. In the Discharge state, the changes in the electrolyte gravity happen when free Sulfate (SO⁻⁴) ions are drawn into the pores of the battery electrodes from the sulfuric acid electrolyte; this leads to a reduction of the specific gravity of the electrolyte with an increase in the water concentration. In the charge condition, the reversible reaction occurs, where the sulfate ions are passed out from the battery electrodes back into the electrolyte and increase the gravity of the electrolyte. During the above-said discharge process, the sulfuric acid is used up, and Lead Sulfate is formed on the battery electrodes. This is the natural

process within the battery during discharge. Battery recharging is necessary to avoid hard sulfation or conversion of lead sulfate crystals into the sulfuric acid electrolyte. If irregular recharges are done, the hard lead sulfate crystals will continue to develop and become difficult to break down at recharging conditions. This will lead to a progressive failure, which then the battery becomes an unusable mode. Many other dendrites (detrimental) actions occur on the surface of the battery electrode while in a discharged condition. The corrosion effect on the lead grids and the connectors increases with the reduction of the electrolyte specific gravity. State of charge during long periods of storage time must be checked and maintained as per the manufacturer's instructions. Several other conditions that can lead to battery failure are inadequate charging or having the battery in idle/inoperative condition for prolonged duration in the vehicle. Overcharging and internal battery temperature should be avoided to ensure optimum life. The addition of distilled water externally is required to have a long life of the battery. At longer charging, the failure of effective Hydrogen and oxygen recombination leads to Water loss, which is typical in lead-acid batteries. This process is also termed as electrolysis as the evaporation of Hydrogen takes place. Lower electrolyte levels or unfilled space which exposes the lead electrode in operation will result in quick failure of the battery immediate with capacity loss and other to follow. LAB requires to maintain the electrolyte levels over the top of the electrodes /above the minimum marking lines mentioned on the outer plastic case of the battery. VRLA/AGM (Valve-Regulated Lead-Acid / Absorbed Glass Mat) battery has a fixed quantity of electrolyte added at the manufacturer. Once the battery is permanently sealed and charged, it must be never opened until its end life, the opening of the LAB is

restricted due to safety and environmental conditions. Currently, AGM technology holds the market demand. Based on AGM (Absorbed Glass Mat) technology, the batteries offer good electrical performance characteristics when compared to flooded technology. The AGM batteries are used in a wide range of applications such as Power backup, Electric/Hybrid Vehicles, Traction, Emergency Lighting, et cetera.



Fig. 1.1 Components of Lead Acid Battery.

1.1.1 Lead-Acid energy storage Construction

Six Lead-acid (Galvanic) Cells are interconnected in a series fashion to obtain a typical 12-volt lead-acid battery. Where these cells are enclosed in a plastic container with six compartments, each cell comprises of two different types of electrodes one being the anode (negative) lead (Pb) electrode and the other being the cathode (positive) lead dioxide (PbO₂) electrode. These two electrodes are separated by an insulating material, either AGM or Polyethylene (PE) Separator.

a) Electrodes

A typical lead-acid battery electrode contains Lead Oxide (PbO) as active material, used in the reduction-Oxidation reaction, and a current collector grid made of Lead (Pb) Alloy, as shown in fig. 1.2. As the metallic lead (Pb) is malleable, alloying additives like Tin (Sn), calcium (Ca), antimony (Sb), et cetera, are used to create lead alloys to enhance the electrochemical properties of the battery. The combination of the Lead alloy grid and the lead oxide Active Material (AM) form a plate/electrode. After the formation process, the lead oxide in the positive/cathode (+Ve) electrode becomes lead dioxide and Lead/spongy lead at the negative/anode (-Ve) electrode.



Fig. 1.2 Battery Current Collector.

b) Electrolyte

As mentioned earlier, the electrolyte is a diluted Sulfuric Acid (H_2SO_4) which is used in the lead-acid battery. The Concentrations of acid are varied with the designs, but generally 30 to 40% of weight is used in the battery. Generally, electrolyte exists as positively charged hydrogen ions (H^+) and negatively charged sulfate ions (SO⁻⁴), which acts as a crucial ingredient in redox reactions. In other designs, treated silica (SiO₂) along with other additives are added to the electrolyte to create a gel from and termed as Gel batteries. In AGM batteries the electrolyte can be trapped and can be used as required, with this design, there are several advantages, like the electrode cells can be mounted in any orientation without any maintenance. In other designs, distilled water needs to be added for prolonged life.

c) Separator

The primary role is to insulate the positive and negative electrodes. The separator is of a porous membrane that prevents the dendrites by reducing the shedding of the active material, causing a short circuit. Majorly, there are two types of separators in lead-acid designs: microporous membrane made up of polyethylene and other absorbed glass mats (AGM) by silica. The microporous membrane present allows the free flowing of electrolytes in flooded form. However, the AGM consists of glass fiber and is soaked in the electrolyte; the fiber absorbs the acid and tends to sink into the solution. Therefore, the advantage of AGM when submerged in the electrolyte solution is that it provides the stability of avoiding electrolyte spills and stratification.

As discussed earlier, the Lead-Acid Battery consists of the following components: we can see in the below Fig. 1.3



Figure 1.3 Lead Acid Battery Construction (Image source: www.circuitdigest.com)

The components of a Lead Acid Battery are Positive and Negative Electrode/ Plates, Separator, and Electrolyte enclosed in a rigid plastic container. A compartment in the plastic container holds a 2V design with the above-said components. Due to this, a series of 2V is connected to make a 12V lead-acid battery (6 x 2V/cell = 12V).

Although lithium-ion batteries seem to grab most of the headlines these days, lead-acid has long been the go-to chemistry for renewable energy storage. Despite their size and weight, lead-acid batteries can deliver enormous amounts of power more cost-effectively than any other energy storage.

Tomorrow's electric grid may require higher energy storage to incorporate more copious quantities of intermittent renewable energy resources. In recent times, LAB has been used as backup power in off-grid applications. Certain limitations in the cycling for these applications need to be addressed as the battery cycles from a charged to the discharged state and then back to a fully charged state, the lead and lead dioxide react with the sulfuric acid electrolyte in the battery, which produces the lead sulfates in the electrodes, known an as sulfation. Over a period, these crystal sulfates accumulate on the negative electrodes and hinder the reversible reactions. Therefore, the battery is limited to perform at different cycle rates.

1.1.2 Electrochemical Reactions of LAB at different stages:

a) Initial Reactions: At electrode preparation where the active material basically lead oxide (PbO) is mixed with sulfuric acid along with binder (Polyester). And carried out with curing process. The application of the batteries requires the either 3BS/4BS crystals in the electrode for effective formation and higher active material utilization. 3BS is required for automotive application and 4BS for stationary application.

$$PbO (AM) + H_2SO_4 = PbSO_4 + H_2O$$

$$\tag{1}$$

Curing Environment @ $<70^{\circ}$ C: PbSO₄ + 3PbO = 3PbO \cdot PbSO₄ (3BS) (2)

Curing Environment @ $>70^{\circ}C$: PbSO₄ + 4PbO = 4PbO \cdot PbSO₄ (4BS) (3)

b) Formation of the cells: At Finishing Stage the formed lead sulphates (PbSO4) is further converted to lead dioxide (PbO2) in the positive electrode and lead (Pb) in the negative electrode.

+Ve Electrode:
$$PbSO_4 + 2H_2O = PbO_2 + H_2SO_4 + 2e^- + 2H^+$$
 (4)

-Ve Electrode: $PbSO_4 + 2e^- + 2H^+ = Pb + H_2SO_4$ (5)

c) Discharging/Charging: At indiviaul electrodes the electrochemical reations takes place at operation as shown in the Fig 1.4 giving out a 2.04 potential difference across the electrodes.

-Ve Half Cycle: Pb +
$$H_2SO_4 \leftrightarrow PbSO_4 + 2H^+ + 2e^-$$
 (E⁰ = ~0.356 eV) (6)
+Ve Half Cycle: PbO₂ + $H_2SO_4 + 2H^+ + 2e^- \leftrightarrow PbSO_4 + 2H_2O$ (E⁰ = ~1.685 eV) (7)

Both the half cycles contribute to the total voltage of Eq. (1) + (2): 2.04eV



Fig. 1.4 Operation of Lead-acid battery (a) Discharging (b) Charging.

1.1.3 Current challenges for Lead Acid Battery

Apart from manufacturing flaws, emerging needs like Hybrid/Electric Vehicle demands higher current charge and discharges, leading to a development of advanced lead-acid battery performance to suit the requirements. Due to this, the battery is stressed out in impacting the performance. The two major areas that hinder the emerging requirements are:

- 1. **Premature failure of the battery:** Where the development of high resistive film between the lead grid and the active material of the positive (Cathode) grid hindering the performance of the battery, specially at high-current discharges. This further leads to grid corrosion and permanently damage the battery performance.
- 2. **Progressive Failure of the battery:** This is where the rapid sulfation growth happens in the negative electrode, leading to irreversible hard sulfates formation withing the active material and on the surface of the electrode, at rapid continuous charge/discharge condition.

Generally, flooded/AGM batteries are used for Start Light Ignition (SLI) as one of the applications in the automotive industry. As the demand for higher performances evolves, the possibilities of failure modes also increase. As mentioned above, the main area of failures is the progressive sulfation and the development high resistive film interface with the lead grid (Premature failure). Targeting resistive film and the sulfation of the electrode to overcome the barriers is by the addition of carbon in nano form.

Recently several industries are studying the use of low dimensional carbons, especially carbon nanomaterials, as additives in lead-acid batteries and advantages on charging and discharging capacities. The studies include the effect of carbon particle sizes in the porous active material on cycling capabilities at different state of charge (SoC) conditions.

1.1.4 Nanotechnology – A possible solution for Energy Storages

Nanotechnology is the study of materials whose dimensions (at least one) range from 1nm to 100nm. This study of microscopic dimensions and tolerances (less than 100nm) can be applied in all other science fields such as biology, chemistry, medicine, material science, and engineering. Nanomaterials are highly reactive due to the high surface-to-volume ratio. They behave completely differently from their bulk counterparts. They show improved properties (electrical, mechanical, thermal, etc.). Using nanocomposites (at least one material is in the nano range) can enhance properties and reduce the size (portability). Carbon is a vital material system with different allotropes. Graphite and diamond are the commonly known allotropes. Recently other allotropes of low dimension carbon (like Fullerene, a nano form of carbon) are discovered like 0D, 1D, 2D & 3D, as shown in Fig. 1.2. These carbon nanomaterials with improved properties are promising materials for energy storage applications. New-age carbon nanomaterials include graphene, carbon nanotubes, fullerenes, etc., among which graphene and its related forms have gained considerable popularity owing to their unique properties.

a) Carbon Materials as Lead-Acid additive:

As said, there is an increasing demand for high-performance capability with respect to the automobile industry. Now the lead-acid battery industries are in search of exploring carbon nanomaterials for enhancing the performances of the batteries. Carbon materials are used to lay a conductive network in Lead-acid batteries to suppress sulfation occurring on the surface of the negative electrodes at high discharge capacity. Due to sulfation, the life of the batteries is limited to short durations, impacting the lead-acid battery industries. The inhibition process by the carbon relies on the size of the sulfates, where the large sulfates formed during high discharges are not reduced back to lead metallic constituents. There have been some attempts to reduce and restrict the size of the sulfate crystals by using nanocarbons. This has led to maintaining the PbSO₄ crystal size as minimum as possible by using of low dimension carbon materials. Various Carbon allotropes have been explored by different research groups for increasing the specific energy and the performance of charge acceptance & high discharge studies as per Japanese international Standards (JIS). The low cycle life, water loss & selfdischarges of the lead-acid batteries is due to the lack of purity in carbons resulting low recombination rate of hydrogen onto the negative electrodes. The high structured carbon nanomaterial as additives would help the transportation of ions in the electrode and enhance the active material utilization (specific energy) for higher cycles.

Researchers find carbon to be incorporated into the lead-acid design in various configurations. One technique is to blend the carbon additives into the lead oxide paste, making the negative electrodes. Alternatively, the regular lead grids used for pasting can be replaced with colloidal carbon on the surface of the lead grids before being taken to the pasting. This hybrid design dramatically improves cycle life over traditional lead-acid batteries.

b) Graphene and its importance

Out of all nanocarbons, Graphene is the most suitable material because it is a 2-Dimensional material which can accommodate maximum interaction with the active material in electrode preparation of LAB. And holds the unique characteristics due to its preparation in cost effective manner. Graphene is a two-dimensional layered material where carbon atoms are bonded with sp² configuration in a hexagonal closed packed system (HCP), which belongs to graphite family. The figure 1.5 shows that the graphite allotropes of carbon atoms packed densely in a honeycomb lattice. Since its discovery, which lead to Nobel prize in 2004, graphene has attracted much attention from scientists due to its unique properties. Graphene is the most durable material ever produced, with superior thermal and electrical properties. It indicates that it is suitable in potential applications like electronic devices, aerospace applications, automobiles, etc.

Single layer Graphene material has a large theoretical surface area of ~2630 m²g⁻¹, exceeding graphite surface area (<10 m²g⁻¹). This higher surface area compared to other materials makes it possible to store more charges which can improve the performance of the energy storages.



Fig. 1.5 Carbon Allotropes:- (0D) Fullerene (1D) Nanotube (2D) Graphene (3D) Few layer graphene (FLG) or Graphite [Image reproduced from Wan et al. (2012) with permission of American Chemical Society]

The below properties which the graphene material is set apart from other

materials and makes it unique are:

- Theoretical specific surface area (2360 m^2/g)
- Electron mobility (200,000 cm²v⁻¹s⁻¹)
- Mechanical Strength: The Young's modulus (~1.0 TPa)
- Thermal conductivity (3000 5000Wm⁻¹K⁻¹)
- Optical transmittance (~97.7%)

Synthesizing 2D carbon material (Graphene) has many routes, which are listed

below, and several other facile routes are also emerging.

- Mechanical Peel-off method
- Atomic epitaxial growth on metal substrates
- Pyrolysis of sodium ethoxide
- Unzipping of carbon nanotubes
- Graphite delamination process

• Chemical Vapor Deposition (CVD), et cetera.

Due to its unique property's graphene is studied in several applications other than energy storage are the few identified areas below:

- Gas detection
- Graphene as transistors
- Touch Screen Transparent Conducting
- Heat transfer systems
- PV solar cells.

1.2 Objective of the study

Our objective is to address the limitation of traditional lead-acid energy storage and provide a solution by rational structure design according to the below principles and criteria. This aims to develop high-performance lead acid devices for emerging applications for Hybrid/Electric automotive and On/Off-grid storages. The goal is to fundamentally understand the operational mechanism and improve the performances like Specific capacity (Ah/g), charge acceptance, High current discharge & life cycle of the lead-acid batteries. Generally, this proposed work is dedicated to the following aspects:

1) Understand and explore the mechanism/behavior of the materials of leadacid energy storage.

2) Fundamentally and technically find the bottlenecks of current carbon materials for Lead-acid energy storage.

3) Develop and implement the novel synthetic method to obtain a new class of carbon materials aiming at its current limitations.

4) Fabrication and integration of low dimensional carbon materials into the

battery component and examine their electrical performances.

1.3 Scope of the Present Work

The scope is to improve the high current discharges and charges of the battery. These benefits are achieved by incorporating carbon nanomaterials on the lead grid and in the electrode's active material. This eliminates/reduces the development of the resistive layer and the sulfation in the electrode. This technique could increase the conversion efficiency of the battery in automobile vehicles by significantly increasing the performance of the batteries which need to provide adequate power.

Moreover, it is to increase the cycle life of a battery by using nanomaterials for making the electrolyte availability in the battery from forming the sulfation on the solid electrodes. This also prevents the low-level discharge in a conventional battery, which dramatically increases the battery's shelf life.

Lead Acid batteries with electrodes made from nanostructured carbon significantly improve the cranking ability even at low temperatures.

Batteries using graphene sheets to store electrical charges improve lead-acid batteries' capacity and battery lifetime.

1.4 Organization of the Thesis

Chapter 2 provides a review of the previously published research on Graphene – A carbon-based nanomaterial Both synthesis and application investigations for lead-acid energy storage systems are presented.

Chapter 3 discusses the identification of gaps in the available works of literature and summarizes the scope of the present work.

Chapter 4 describes the methods and the preparation of carbon-based nanomaterials.

Chapter 5 is devoted to summarizing the characterization methods of the prepared material and analyzing the process parameters.

Chapter 6 describes the present investigation, in which an attempt is made to prepare the desired electrode and analyze the electrode behavior before making it into a battery, maintaining the quality protocols as per the industrial standards.

Chapter 7 presented the results and discussed which describes the improvement in the electrical performance of the lead-acid batteries due to the application of carbon-based nanomaterial.

Chapter 8 presents the summary of the research and the major conclusions drawn from these investigations. Further research needs are also identified and presented.

17

<u>CHAPTER – 2</u> <u>Literature Review</u>

In this chapter, we will be discussing the works carried out by various research groups in developing advanced lead-acid batteries with carbon materials and the synthesis of 2D carbon nanomaterial with several approaches are discussed. The first part of the chapter discusses about the exploration of graphene synthesis and followed with the key findings of carbon usage in lead acid battery. The summary and the identified gaps were collated and presented.

2.1 Forms of Carbon

As per Sandia National Laboratories, USA report (SAND2011-3459, 2011), the carbon is classified in to two categories as Amorphous and crystalline carbon as shown in Fig. 2.1.

- 1. Amorphous Carbon (Carbon Black, acetylene black, furnace/Channel/Thermal black & lamp black)
- Graphitic Carbon (Graphite, Expanded/Exfoliated Graphite, Graphene, Fullerene & Nanotubes)



Fig 2.5 (a)X-Ray spectra of Crystalline to amorphous carbon grades – Sandia National Laboratories, USA (2011).

However, the advantages of crystalline carbon are more when compared to amorphous carbons. Due to their structural stability these materials gained attention over the years by several research groups. While the amorphous carbons are heavily produced by oil industries and the crystalline materials like graphite are naturally found and most crystalline carbons are synthetically grown/made. Out of all crystalline carbons graphene has been predominantly investigated due to its unique properties as discussed in earlier chapter. Now, let us investigate the synthesis procedures and the challenges faced by several scientists in making graphene material from graphite.
2.2 Synthesis of Carbon Nanomaterials - For Energy storages

Stankovich et al (2007) had synthesized graphene by reducing the graphite oxide chemically, following colloidal suspension method. The prepared material was characterized by SEM, XRD, Raman, EDAX, Conductivity etc., and confirmed. Further studied by incorporating as conductive filler in the composites for energy storage device.

Kaniyoor et al (2010) had used hydrogen atmosphere to synthesize graphene by exfoliating graphite at low temperatures (<100°C). Later characterized using XRD, SEM, Ram, XPS and N₂ adsorption techniques for physical and chemical properties. **Marcano et al (2010)** extracted graphene by oxidizing the graphite flakes with KMnO₄ and other acids like H₂SO₄ and H₃PO₄. He also compared with other routes and disclosed that his method leveraged in less defects with high productivity (50% Yield).

Hernandez et al (2009) adopted liquid phase exfoliation by inducing mechanical forces of graphite solution. Graphite is mixed with a surfactant and stirred in water overnight. This solution is later sonicated, centrifuged, filtered and dried to obtain graphene material. Various characterization techniques like microscopic and spectroscopies were followed the formation of graphene material.

Chen et al (2009) has also adopted H_2O_2 as strong oxidizer along with KMnO₄ and H_2SO_4 acid allowing the reactions to take place under 20°C. Later stirred at 35°C and washed with HCL solution before drying. The obtained dried powder is then characterized and found to be acceptable, but with more time duration of >2days.

Chung et al (1987) revealed different ways of exfoliating graphite to obtain graphene nanostructures. Three major ways are highlighted by the scientist 1. By

using oxidizers and acids creating a exothermic reaction. 2. Subjecting the acid treated graphite to external heating. 3. Allowing the acid treated graphite to mechanical forces like sonication.

G. Venkata Ramana et al (2011) has successfully prepared graphene/PANI composite with high thermal stability and good electrical conductivity for energy application, which found to be reliable.

Zhihong Luo et al (2013) had prepared Graphene/PANI composite for supercapacitor application. The graphene is prepared by reflux method in combination with aniline oxidized by ammonium persulfate (APS). The composite have better capacitance than individual, enabling as supercapacitor electrode.

Srinivasulu Reddy K et al (2015) reported the usage of graphene as metal matrix composites enhances the mechanical and thermal behavior of the composite. A powder metallurgy process has been adapted to make graphene /aluminum metal matrix nano composites (AMMnCs) with enhanced properties.

2.3 Application of Carbon in Lead Acid Battery for progressive failure

Although the foundations laid for the primary batteries (Non-Rechargeable) by Luigi Galvani in 1780's the wide spread of the secondary batteries (Rechargeable) was not seen until Raymond Gaston Plante inventing the lead acid cell in 1859 later named as Plante battery on his memory. Later, several inventions & developments has been made in terms of operations/usage.

The plante's battery consisted of nine cells in parallel and was constructed by winding to led strips with a rubber strip and a heavy lenin or felt material is placed between them s shown in Fig. 2.2. This arrangement was housed inside a

21

cylindrical jar after filling the cells with dilute sulphuric acid the plates were found by using bunsen battery this is akin to plante formation.

The popularity off plant battery increased with the development of electrochemical dynamic by W. V. Siemens et. al in 1866 and with the invention of Magneto electric generator by Societe Gramme et. al around 1873 as cheap reliable and practical sources of direct current for battery charging.

Twenty years after plante discovery of lead acid system Gladstone and Tribe et. al explained in 1881 and 1882 the reactions occurring in the system on the basis of "double sulphate theory" meaning that both lead dioxide and lead were converted into lead sulphates during discharge also this theory met with violent opposition, and it was confirmed by several others.

Around 1919 G Shimadzu et. al in Japan device the forerunner for the present-day ball mills for making lead oxide powder as active material. From 1920 onwards newer applications like emergency power supply air conditioning of rail cars and a host of other services on ships aircraft buses and trucks began to play important role in enhancing the market share of lead acid batteries.

Initial days wood was used as separators. Usage of Expanders began from 1920 and new types of separators were developed to replace the wood separators. When rubber separators were substituted for wood separators the performance of the batteries was not satisfactory and this failure was traced to the lack of wood ingredients in the electrolyte. Soon, finely grounded wood was included in the negative paste composition to obtain superior low-rate high discharge performance. This experiment has let for increased experimentation on the effectiveness of adding additives, which led to the present-day use of carbon, barium sulphate and lignin

22

derivatives.

Breakthrough in the development of commercially viable sealed lead acid battery came after the invention of sealed nickel cadmium cells by Lange et. al. He was the first to outline the principle of oxygen cycle happing by the electrodes in his US patent 1938 then later in 1950s and early 1960s the strong foundations came necessary for the commercial development of sealed batteries. Simultaneously two technologies on one based on gel electrolyte and other absorbed glass mat was developed. The process of fine refinements in the technology is continuing one till date making improvements in several fields of battery evolution as well as in application areas.



Fig. 2.2 a) Schematic of Plante's Lead-Acid Cell. b) Early illustration of battery with nine cells. (Image Source: www.ciando.com)

Carbon is used in the name of expander in the lead acid battery. The term expander is applied to the mixture of additive materials added to the negative paste either to maintain or enhance the performance of the battery during the operations. Typically, an expander contains barium sulphate, carbon black, lignosulphonate and other organic or inorganic materials in different proportions suitable for various applications. Early studies on the effects of carbon edition in the negative space showed that carbon functions primarily as a conductor thereby enhancing the formation efficiency and reduces the level of residual sulphate, later it was also observed to improve recharging condition particularly under deep discharges. It was further concluded that when combined with barium sulphate and organic component atypical carbon addition of around point to percentage influence on the discharge performance and cycle life of the batteries for different applications shown in below Fig. 2.4, 2.7 and 2.9.

Masaaki Shiomi et al (1998) estimated the usage of carbon black leads to form a conductive network between the active material to enhance the recharging ability especially in the negative electrode of the lead acid battery.

Ohmae et al (2003) has identified that the conductive carbons are very much necessary for NAM to benefit from the progressive failure (Sulphation). Therefore, addition of conductive carbons will reduce the PbSO4 formed crystals effectively and increase the life cycling ability of the batteries.

Spence et al (2008) presented the improvement in battery performance by using carbon in the NAM. The main beneficial effect of using carbon is, it alters the pore structure of NAM and helps electrolyte interaction effectively with the active material.

Boden, et al (2010) described that the carbon particle is suppressed under the lead sulphates and couldn't be able to get the benefit by losing it conductive path. With the increasing carbon concentration, he could be able to achieve superior

performance. However, the surface area and the structure of the carbon impacts the pore structure of the NAM. It is also noted that with increase in carbon concentration the water loss is evident, this is due to the structural nature of carbon used.



Fig. 2.3 SEM image of Negative plate with Carbon Black after life cycle. Carbon Black particles are completely covered with hard $PbSO_4$ particles. (Scale: 20μ)

Kozawa, et al (1999) explored in a novel way by adding carbon to the electrolyte as an additive. While carbon acts as nucleation site for lead deposition during charging and helps to reduce the sulfation in the NAM.

Pavlov, et al (2011) expressed that by having the carbon would enhance the surface area of the plate. Which keeps the plates in active mode by restricting the sulphates growth in the negative electrode.

Moseley, et al (2018) elucidated the carbon as a capacitive component, by increasing the carbon concentration in the negative electrode, the conductivity path has also increased with improving the battery performance. However, these improvements are compromised with the water loss at higher concentration of high surface area carbons. Particular to Hybrid Electric vehicle application the rapid operation of the battery has sustained with the use of carbon.





Fig 2.4 Hybrid EV/Regenerative Braking of Nissan Leaf, where Lead acid battery is used as a secondary source.



Fig. 2.5 Schematic diagram of effect by without and with carbon additives on negative electrode in operation (X. Zou et al., 2015).

Fernandez, et al (2010) explained about the side effects of using high surface area (>500m²/gm) carbons with higher concentrations. The enhancement in charge acceptance had proved that carbon works as a capacitive component in the lead acid battery. However, the water loss has resulted as a limitation subjected to the

form of carbon used. The carbon crating the conductive path has been successful in reducing from the non-convertible PbSO₄.



Fig. 2.6 Restricting the formation of hard sulfates by adding carbon in the NAM (Fernandez et al., 2010).

Spence, et al (2009) & Valenciano, et al (2017) also found the optimal use (0.2 - 2% loading w.r.t active material) of graphite flakes in the Negative electrode and improving the battery performances for Ideal Start Stop (ISS) / PSoC application.



Fig 2.7 Ideal Start-Stop/Cranking available in automotive of 4wheel and 2wheel drive.

Jakub Lach et. al., (2019) has developed a novel Pb/C electrode for stationary backup application at high discharge currents. And compared the performance with the regular batteries and found to be recommended, with Pb/C based battery the charging/discharging cycles were able to achieve 275 cycles, almost 2times of the control batteries. However, the construction of the electrode component involves

more time than making a conventional battery. This restricted to the laboratory and needs to be worked for commercial scale.



Fig. 2.8 End voltage of failed batteries operated with complete charge/discharge cycles. Jakub Lach et. al., (2019).



Fig 2.9 Telecommunication/Solar Integration of lead acid batteries.

2.4 Application of Carbon in Lead Acid Battery for premature failure

Generally, Lead grids are used for preparing electrodes for lead-acid batteries. The lead grids undergo severe stress under electrochemical operation particularly positive grids. These grids corrode in due course of time causing lower battery performances. This was observed with low antimony (%) and no calcium (%) in the lead alloys. But this corrosion had been addressed by several researchers by adding Tin (Sn) with different concentrations to the lead metal as per the application. But very minimal study has been carried on the development of the resistive layer on the grid/paste interface.

Paul Ruetschi (2005) has explained the development of resistive layer with grid/paste interface and the loss of coherence with positive electrode. Below illustration depicts the transformation of Pb/PbO₂ to PbSO₄., which impacts the performance of the battery.



Fig. 2.10 Passivation layer on the positive grid while higher current

discharge.

Patent No. CN104362301 (2017), discloses a preparation method for a carbon coated titanium - based lead dioxide positive plate which is obtained by coating a carbon material on the surface of a metallic titanium mesh (instead of Lead mesh) with a vapor deposition method. There are other methods to improve the grid performance in lead - acid batteries. For example, lead - carbon, including lead-graphene and lead-graphite, composites have been tested as possible positive current collectors for lead - acid batteries. It has been shown that neither graphene nor graphite has participated in the electrochemical process but have improved the corrosion and electrochemical behavior of the composite materials.

Few methods employed in the past, as disclosed in **US10388964**, **2019** patent, prepared a resin solution with PPCP/ABS (Poly - Propylene Copolymer / acrylonitrile butadiene styrene) as the base material and reinforced with graphene to reduce the resistivity of the resin/polymer and this polymer matrix composite is applied on the grid to act as an insulator. And the patent didn't mention anything about the performances of the battery.

Similarly, in **Patent No. CN103811767, 2018** patent a polymer (PTFE -Polytetrafluorethylene / PVDF -Polyvinylidene fluoride) is used as a binding agent and reinforcing with graphene to increase the mechanical strength and electric conductivity of the grids. In the prior art the graphene material is used a reinforced material into the polymer matrix either to reduce the resistivity or to increase the mechanical strength.

However, the prior art hasn't disclosed about the interface (resistive/sulphate layer) on the positive grid. Employing Polymer/ Resin related binding agent as a precursor or as a base material can be only useful as an insulating material. With polymer (binder/resins) applied on the lead grids also increases the grid weight and influence overall weight of the battery.

M. Saravanan et. al., (2015) attempted to explain the benefit with CNT coated negative grids. His work has explained the successful coating (Fig. 2.11) and the enhancement of grid properties with CNT material and suggested for more investigations, which are needed with other designs and materials to resist the piled-up resistance over the grids.

30



Fig. 2.11 EDX spectra (a) Grid without coating (b) Grid with CNT Coating.

Therefore, there remains a challenge and need to address development of the sulphate/resistive film on the positive grid/paste interface to overcome the drawbacks of the prior arts.

2.5 Gaps Identified

a) Synthesis of Graphene (2D- Carbon)

The technologies that move forward our modern society such as automobiles, airplanes, lasers, and portable electronics et al, rely on the carbon-based materials which could supply energy continuously. A facile and novel route of preparing graphene material would be the prime focus. The desired material should have stable structural properties and must be helpful in applying to the energy storage applications. Low dimensional carbons have always been in the limelight for studying in a wide range of fields. For example, CNT's and graphene are emerging as next generation materials with high mechanical and thermal properties. Graphene with adequate surface area and pore structure would show great promise in the applications such as batteries, supercapacitors, and fuel cells.

b) Graphene for Lead Acid Energy Storages

Since the allotropes and forms of Graphitic carbons was established as electrochemical intercalated material for the battery application. Intercalation/deintercalation of hydrogen and sulfate ions, in and out of the porous active material upon cycling forms the requirement of carbon as the most suitable anode of the battery. Carbon, thereafter, becomes the preferred additive for negative active material due to its good cyclability, better charge acceptance and high current discharge characteristics. To date, amorphous carbon black is the most widely used form of carbon in NAM of LAB. Therefore, crystalline carbons like graphene need to be developed towards a higher capacitance and better stability. Generally, carbon materials are used in powder form as an additive in NAM. Now the approach of applying must go beyond the NAM, to get the maximum benefit out of it.

2.6 Summary

In this chapter few conventional graphene synthesis methods are being examined along with an improved and novel approaches. The best suited existing method or by modified method will be adapted to produce graphene in a meek and tranquil manner. Further Graphene is made use as an additive along with BaSO₄ & Lignin for negative electrode of lead acid energy storages and will be tested.

Three major areas need to address for the performance improvement of Lead Acid batteries.

a) Synthesis of 2D-Graphene.

32

b) Development of graphene colloidal solution for coating the lead grids.

c) Preparation of Negative electrode with graphene as an additive.

Above to these areas, developing high-performance electrodes and devices should follow the below criteria:

1) Active interface between grid and the active material.

2) Electrically conductive pathway in the NAM.

3) Mechanically stable electrode to sustain the vibrations caused in the field operation.

To achieve the criteria mentioned above, the design of high-performance carbons is supposed to follow these directions:

a) Well-defined carbon structure and pore configuration.

To achieve a high energy/power density, LAB requires the carbon to efficiently intercalate Hydrogen and Sulphate ions on to the electrode surface to accumulate as much as possible, thus a large surface area and approachable surface/bulk interiors is desired when designing and synthesizing carbons. From another aspect, to obtain a high-power density, the carbons are demanded to transport ions and electrons efficiently. Therefore, a good conductive framework including interconnected pores and appropriate pore size are desired.

b) Electrochemically active and stable interface

All the chemical reactions take place at the interfaces of Electrode - Electrolyte and Grid - Active material. Hence, foremost, the carbons should be well approached and infiltrative by electrolyte. The hydrophobic/hydrophilic properties are influenced by the nature of carbon materials, carbon surface functionalities, and carbon surface charge properties. Moreover, during the electrochemical charging/discharging process, carbon should be chemically/mechanically stable in electrolyte in the given electronic conditions. The carbon with less/no side reactions, tolerant voltage/current ranges, and robust frameworks are desired to achieve this goal.

<u>CHAPTER – 3</u> PROBLEM DEFINITION

In the previous chapter a brief explanation about the preparation of Carbon/Graphene and its applications in lead acid battery were discussed with possible areas for future research. Further there is much scope in developing the advance lead acid batteries with graphene material. In this chapter we will identify the areas hindering the performance of the battery the failures related to lead acid battery.

3.1 Statement of The Problem

To enhance the performance of lead acid energy storage like the initial discharge capacity / energy density, charge acceptance, high current discharges, and cycle life. The material studies need to be carried out, to provide better understanding of the material properties. Taking forward with electrode preparations to enhance the electrical properties of the lead grid and the active material of the battery. To achieve this goal, graphene material is explored as an additive material in the electrode preparation to enhance the electrical performances of the proposed research. In below we will discuss about the failures associated with lead acid battery.

3.2 Reasons effecting Lead Acid battery performance

The reasons mentioned below are categorized as per the device fabrication and operation.

- a) Catastrophic Failure (CTF)
- b) Premature Failure (PMF)
- c) Progressive Failure (PGF)

a) Catastrophic Failure (CTF)

This results, either from manufacturing perspective or from the operational conditions at the time of preparing the energy storage devices.

- 1. Mismatch of the component selection (+Ve/-Ve electrodes)
- 2. Poor manufacturing quality (Interconnections/defect separators etc.,)
- 3. Under charging & Over charging of the device
- 4. Internal or external damage of the cells (Terminal leakages / Heat seal of PPCP material)

The above issues can be addressed and fixed by controlling the process parameters & manufacturing feasibility with quality controls.

b) Premature Failure (PMF)

Based on the electrical characteristics of lead acid energy storage the main reason of premature failure, is the loss observed at deep discharges of the lead alloys with the development of a high resistance film between the lead grid and the active material (Grid/Active Material Interface) shown in the fig. 3.1, when operated in Over/partial charging / Idle condition. Even at the time of active material pasting process (electrode preparation) sulphate/resistive layer is developed due to the interaction of H_2SO_4 available in the paste. But, at the operations providing a constant/abnormal charge voltage, will increases the development of passivation layer on the grid and could not be able to perform at high current discharges with required capacity.



Fig. 3.1 Step wise formation of Passivation/Resistive layer on the lead grid. (This interface hinders the performances of the battery by resisting the flow of current)

c) Progressive Failure (PGF)

Progressive failure (PGF) happens with the active material of the negative electrode. As gradual growth of sulphation occurs at the negative electrode of the battery when subjected to higher number of cycles of charging & discharging or prolonged storage. A complete discharge battery will allow the sulphation (PbSO4) to pileup as shown in fig. 3.2. The application associated with this process are Hybrid/Full EV, PSoC and HRPSoC conditions. This sulfation will also affect the Initial discharge/Energy density, charge acceptance and High current discharge abilities. These sulphated crystals would increase the resistance of the electrode, which wouldn't be reconverted back to metallic lead. This would also require large currents to dissolve these sulphate crystals but would not be successful due to the lack of conductive path in the electrode, which are suppressed by the sulphates. In will also form dendrites on the electrode/separator interface, causing short circuits of the electrodes and permanently damaging the battery.



Fig. 3.2 Formation of Sulphation on the negative electrode over a period.

The specific purpose of this proposed research is to enhance the performance of lead acid energy storages by addressing PMF and PGF. The performance enhancement can be analyzed with the parameters such as high current discharges, charge acceptance, Initial discharge capacity/energy density and life of the device.

3.3 Our Approach for Possible Solutions

- 1. For PMF, lead grids are coated with prepared Graphene/Carbon colloidal Solution to limit the formation of passivation layer on the grids.
- 2. For PGF, Graphene/Carbon is added as an additive to inhibit sulphation in Negative active material.

3.4 Experimental Approach for Premature Failure (PMF)

Step 1: Graphene Colloidal Solution Preparation.

- Step 2: Coating the grids with graphene solution on the lead grids.
- Step 3: Evaluation of coated electrodes.

3.5 Experimental Approach for Progressive Failure (PGF)

Step 1: Negative Electrode preparation with Graphene as an additive in the active material. The advantage of having Graphene in the active material is illustrated in Fig. 3.3.

Step 2: Evaluation of assembled graphene aided electrodes as a device.



Fig 3.3 Mechanism of graphene in active material

Herein, this proposal presents a facile synthesis route of graphene preparation to achieve a unique property with high productivity. There are several advantages and disadvantages of synthesizing 2D-Carbons/ Graphene material in each way. Graphene prepared by Chemical Vapor deposition (CVD) or mechanically exfoliated technique have the purest with less defects. It holds the properties near to the theoretical values. But the cost with these techniques is high and the rate of production is less which is not a cost-effective approach. Whereas other methods involving chemicals like hummers method can produce graphene in large scale but lacks in purity (more defects and presence of other functional groups). So, there is a need for a better method with reasonable purity and productivity. In this work few synthesis methods presently in use are examined along with an improved and novel approach to prepare graphene by simple means. The prepared graphene is taken forward to address Premature Failure (PMF) and Progressive Failure (PGF) as a coating material on to the grids and as an additive in active material of negative electrode for lead acid energy storages. The electrochemical measurements are carried out in standard sulfuric acid as an electrolyte for both the approaches.

3.5 Summary

From the previous chapters we have been able to Identify the areas for improving the battery performances like Charge acceptance, Initial Discharge Capacity/Energy density, High current discharges. PMF & PGF, are within our scope, which affects the performances of the battery. To address the above said failures, graphene material has been chosen, due to its unique electrical properties. Application of graphene material in colloidal phase for targeting PMF. And the same has been applied as an additive to the Negative active material for limiting the PGF.

<u>Chapter 4</u> <u>Nanomaterial Synthesis</u>

From the previous chapters we understood how crucial the preparation of graphene is. In this chapter we will be preparing the graphene material in a facile and cost-effective way. And we also characterize the prepared graphene material with Microscopic, spectroscopic etc., related qualitative tools.

As we discussed about graphene in earlier chapters, as a promising material which can be used in many fields (electronics devices, gas sensors etc.). But to use any material for commercial application there should be an industry level cost effective synthesis method. In this chapter few synthesis methods are presented to corelate and examined along with an improved and novel approach.

4.1 Synthesis of Carbon Nanomaterials- Existing Methods

- Mechanical Exfoliation (Scotch Tape)
- Chemical Vapor Deposition
- Hummers Method
- Liquid Phase Exfoliation

4.1.1 Mechanical Exfoliation (Scotch Tape)

In year 2004, Nobel Prize has been awarded for the discovery of 2-Dimensional carbon, which is graphene for the physicists Prof. Andre Geim and Prof. Konstantin Novoselov from University of Manchester, UK. They have used a scotch tape to extract graphene from the graphite and were successful in pulling out few layers of graphene, which later examined under an optical microscope (Fig. 4.1) categorized by light absorption.



Fig. 4.1 Mechanically exfoliated graphene on SiO₂ imaged with white light using an optical microscope (Photo credit: Nathaniel Safron).

4.1.2 Chemical Vapor Deposition (CVD)

CVD process is a way of thin film technique where the atomic deposition takes place by the gas molecules (Hydrocarbons - CH₄) as a precursor onto a substrate in layered manner. The rate of deposition, number of layers and thickness of the material can be altered according to our requirements. Several researchers have synthesized with this technique with different temperatures and environmental conditions created in the chamber (Low Pressure CVD, Plasma enhanced CVD, Ultra-High Vacuum CVD etc.,). The process involved with CVD is shown in Fig. 4.2. The benefit with this deposition process is that we achieve in high purity graphene layers, but with low quantity. The other disadvantages with are process is the by products like Carbon Monoxide (CO) which need to be disposed carefully. This method would be more viable for making semiconductors to carryout lithographic steps and related optoelectronics industries.



Fig. 4.2 Steps involved by CVD process:

- (a) Diffusion in process of hydrocarbons (reactant gas).
- (b) Atomic adsorption (Carbon) with the substrate.
- (c) Chemical reactions on the substrate.
- (d) Desorption of unabsorbed carbons.
- (e) Diffusion out process of unabsorbed carbons and by-products.

4.1.3 Hummers method

2:1 ratio of graphite flakes (3gms) and sodium nitrate (1.5gms) is well mixed and Conc. Sulfuric acid (70ml) is added to the mixture and placed in an ice bath to restrict the temperature growth. Later KMnO₄ (6gms) is added and stirred by maintaining the temperature under 20°C. After removing ice bath, the mixture was stirred at 35°C for 30 min. The mixture started becoming thick after 20 min of stirring. A very thick paste was formed after 30 min. DI water (138 ml) was added slowly to the thick paste and stirred to stop the reaction. After 15 mins further addition of water (450ml) to the solution is added to completely stop the reaction and stirred for 30 mins. Later, Hydrogen Peroxide (H₂O₂) with 30% concentration is added 5 ml to reduce metal oxide particles, to wash out easily with water. Change in the color of the solution from reddish brown to brilliant yellow is an indication of reduction of metals in the solution. After reduction of metals, the solution is washed and filtered, later allowed to dry at 105°C for 24Hrs. The dried-out material is the graphene with few layered structures. The whole process takes time about 2days and involves exothermic and violent chemicals.



Fig. 4.3 Synthesis of graphene oxide using hummer's method (Top) Flowchart and (Bottom) experimental pictures

S5

S6

4.1.4 Liquid phase exfoliation

S7

Graphene can be produced directly from graphite using this method. The energy required to separate graphite layers will be produced from the interactions of graphite surface and the solvent used. This process is possible only with few organic solvents like NMP (N-Methyl-2-pyrrolidone), DMF (Dimethylformamide) etc.

Graphite (50mg) is added to DMF (250ml) and sonicated for 2 hours using ultra probe sonication. Sonication can be continued until a clear solution is formed and there should be no agglomeration at the bottom or on the top. The solvent is made evaporated, and the resultant powder is collected after properly drying in hot air Owen for 2hours.



Fig. 4.4 LPE (A) Schematic diagram and (B) Experimental pictures and (C) Graphical representation.

4.2 Our Approach: Microwave Irradiation (Novelty in this work)

In this work graphite is intercalated with oxygen functional groups using aqua regime (3:1). Graphene oxide can be reduced by supplying enough energy in any form (thermal, microwave radiation, solar radiation etc.). When energy is applied water molecules and oxygen groups tries to escape (evaporate) from GO. In this process they break the bonds between individual layers and separate them to form reduced graphene oxide. We cannot remove oxygen groups completely, but after some point it will be stable and holds similar properties to that of pristine graphene. This process is explained below in detail. Materials used in this work: graphite flakes (150 μ m) (Sigma Aldrich). Sulfuric acid (98%), Nitric acid (70%), hydrochloric acid and Hydrogen peroxide were purchased from Merck.

In a beaker of 1 liter, 400ml of acid mixture (aqua regime) in a known ratio was taken and 3g graphite was added to it. It was then stirred for a known time. The reaction was then stopped, and the suspension was allowed to settle down. The clear solution or the acid mixture (supernatant) is then decanted through which roughly 75% of the acid was removed. Dilution with DI water and removal of supernatant was repeated until the pH reduces to 7. It was then dried at 50°C in Owen for 8 hours. Remaining solid is collected and stored. We call it acid treated graphite (ATGF). Microwave treatment of above prepared acid treated graphite expands and becomes less dense as the inter planar distance between the graphite layers increases. This expanded ATG is called graphene warms (GW). Graphene warms are dispersed in ethanol in 0.05 mg /10 ml ratio and sonication is done for 30 minutes using ultra probe sonication. A stable colloidal dispersion is formed, which then dried to collect the remaining solid product.



Figure 4.5 Optimized novel approach used in this work (A) Flowchart (B) Experimental pictures (a) Acid treatment (b) ATGF (c) MWGW (d) MWG **4.3 Summary**

Various methods for synthesizing graphene were studied and initiated novel approach to synthesis graphene in environmentally friendly way with cost effectiveness. Initially using aqua regime graphite was partially oxidized which then exposed to microwave radiation to get graphene worms. By sonicating these worms in ethanol environment until the formation of a uniform solution is achieved followed by evaporation of solvent to collect Graphene material. Instead of exposing acid treated graphite flakes to microwave radiation we can also use other means (Thermal & Solar) to produce graphene which is safe and freely available. Use of other energy sources to realize further effective and cheaper method is continuation and future aspects of this work.

Lack of bulk production of mono layered graphene by chemical routes is a major stumbling block related to this research. Graphene produced from the above said process, would not be much useful in semiconductor related industries producing IC's/ Transistors etc. But the graphene material obtained from the above said process is quite compatible for energy storage industries. The material characterization and application of graphene in lead acid energy storages will be discussed in next chapters.

<u>Chapter 5</u> <u>Material Characterization</u>

In the above chapter, we have discussed about the synthesis of graphene material. Now we will discuss the qualitative analysis of the synthesized graphene with different characterization techniques. And the summary is drawn about the successful synthesis of the desired material, which will be applied to making leadacid energy storage device.

The morphological studies were carried out by using Scanning Electron Microscope (SEM) model 6010Plus/LV of Jeol-USA and Transmission Electron Microscope (TEM) model FEI Tecnai G2 S-Twin. XRD experiments were carried out using Rigaku Ultima IV, Japan.

The Raman Analysis were carried out with Nd-YAG laser source of WiTec, Germany. XRD and Raman studies were used to understand the structural revelation and then to correlate with morphological observations. Resistance calculations are made using 2-probe method. Surface area measurement of the prepared graphene materials were carried out by Coulter, model 3100, USA.

5.1 Scanning electron microscopy

A SEM is used to study morphology of nanomaterials completely or in part. Electrons generated from an electron gun are accelerated using high electric field and directed towards specimen. During the passage through vacuum primary electrons will be focused by electric field to form a narrow monochromatic beam which hits the specimen and liberates secondary electrons. Secondary electrons liberation depends on angle and velocity at which electron beam is hitting. This signal is amplified and transformed to a video scan-image that can be seen on a display that can be saved and processed further.

Electron micrographs of the graphite flakes (starting material) and graphene are observed from the images that the graphite flakes are thick and stacked type structure (in the order of few μ m) as shown in Figure 5.1. The randomly aggregated thin graphene sheets (few of them stacked together forming independent FLG structures) closely associated with each other and forming a disordered solid. The folded regions of the FLG sheets are found to have an average thickness of few nm. The absence of charging during the imaging indicated that the material was constituted with a network of graphene-based sheets that are electrically conductive. It can also be observed from Fig. 5.1 that FLG sheets have lateral dimensions of at least 1 μ m2. Fig. 5.1 is the evidence that the adopted method (i.e., oxidation of graphite flakes followed by microwave irradiation and subsequent probe sonication) is an optimum one to convert very large number of stacked graphene layers in a cost-effective way.



Fig. 5.1 Secondary electron micrographs of (A) graphite flakes and (B) Expanded Graphite (C) Graphene

5.2 Transmission Electron Microscope (TEM)

Wide beam of electrons generated using an electron gun are accelerated towards a thin specimen to form an image. Working principle of TEM is analogous to a standard microscope. Very thin specimen (sample) is used as the electrons need to penetrate through it. Electrons generated by electron gun are made to travel through the thin sample placed in the middle of the equipment of which some are penetrated, absorbed, and reflected. The sample is placed in the middle of the equipment to capture the black and white image on the screen placed at the bottom.

5.2.1 Sample preparation for TEM

Graphene is added to ethanol (0.1mg/10ml) and sonicated to get a clear

solution. Using a micro pipette, a drop of it is transferred onto TEM grid and let it evaporate. This grid with sample coated as a film is used for TEM studies.



Fig. 5.2 (A) Transmission electron micrograph of few layered graphene, (B) corresponding planar view, (C) corresponding SAED pattern and (D) comparison of x-ray diffractograms of graphite and FLG

The above mentioned FLG sheets are transparent to the electron beam as observed in transmission electron micrographs (Figure 5.2(A)) indicating the presence of few layers along the c-axis in a single structure. Comparison of x-ray diffraction data obtained from FLG with graphite (Figure 5.2(C)) showed a typical (002) and (004) with reduced intensity indicating the formation of graphene through exfoliation. The x-ray diffractogram obtained from FLG also indicated that it is highly ordered. Figure 5.2(B) showed the typical sixfold symmetry as expected for Selective Area Electron Diffraction (SAED) pattern of FLG. This is complemented well with that of XRD data. The regular outer and inner hexagon patterns with varied intensity of the electron diffraction spots as shown in Figure 5.2(B) indicate few layers system that is built-in with A-B type of atomic stacking as in graphite.

5.3 X-Ray diffraction

X-ray diffraction is a unique tool of constructive interference to characterize the chemical composition and the crystallographic structures of the material. In general, interatomic spacing of crystals is around 2 to 3 A^o so the X-rays are the most suitable to study the properties of materials. In works based on the Bragg's law.

$$n\lambda = 2dsin\theta$$
 -- (1)

Where,

d = Atomic layer spacing; θ = Incident angle; n = Order of reflection;

 λ = wavelength of the beam.

X-ray diffracted graphs of all the processed graphite are showed in Figure 5.3. The sharp diffraction (peak) of 002 plane is an indication of stacked layers in few layers graphene.



Fig. 5.3 XRD pattern comparisons

Figure 5.3 compares X-ray diffraction pattern of all the intermediate products in the synthesis of FLG from graphite flakes. Diffraction pattern shifts very slightly during this process. Table 5.1 gives the characteristics (full width half maximum and d-spacing) of all materials from graphite flakes to FLG.

5.4 Raman spectroscopy

Raman spectroscopy is very much helpful to study carbon materials purity. The principle is that when a sample is subjected to the monochromatic light beam, the sample absorbs the photons and experiences an inelastic scattering and then re-emits photons. The frequency of absorbed and re-emitted photons is captured, this is called 'The Raman Effect'. The captured frequencies give us the information of the materials vibrational, rotational, and transitional energies of the molecules present in the material.



Figure 5.4 Raman spectrographs comparisons

S.No	Method	Graphite	Solvent/aci	Graphene	
		(g)	ds	Yield	purity
1	Hummers method	1	23ml	0.3g	poor
2	Improved Hummers	1	50ml	0.5g	good
	method				
3	Microwave	1	100ml +	0.65g	good
			33ml		

Table 5.1 Comparison among the three synthesis methods

Table 5.1 compares the three methods in different aspects like productivity, purity, cost effectiveness is given below. From these comparisons it is evident that the graphene produced using the novel approach optimized in this work is comparatively better in terms of purity, cost effectiveness, productivity, and conductivity. Especially high productivity makes it an attractive method to synthesis graphene commercially. Figure 5.4 compares the Raman spectrographs of the three graphene's. Hummer's graphene is having high intensity d band at around 1350cm⁻¹ is an indication of presence of defects in graphene planar structure. Reasons for defects may include impurities like oxygen groups or simply the absence of carbon atoms. Unlike in hummers method there will be no oxygen groups present in graphene network in the microwave graphene which can be backed with less or absence of defect band in microwave graphene. This is the reason for low resistance or high conductivity which will be discussed in below section.

In Graphene material it is know that G and 2D bands in Raman spectrum is observed at ~1590 cm⁻¹ and ~2685 cm⁻¹ as mentioned in Table 5.1. With the higher graphene layers the G band shift towards lower wavenumber and the 2D band shifts towards higher wavenumber and the shape of the 2D band, deforms from sharp to broad peak.

Bands	Hummers	Modified	Microwave
		Hummers	
G band	1585	1585	1581
D band	1352	1349	1344
2D band	2720	2712	2713
ID	229	219	218
IG	440	259	394

Table 5.2 Bands and intensities of 2D carbons (graphene) by different routes.
I2D	316	224	292
ID/IG	0.52	0.84	0.5
IG/I2D	1.3	1.15	1.34

5.5 Resistance Measurement

Pellets are typically created by compressing an original material to study the resistance of the material as per IS 12178 standards. Usage of pellets instead of powders to study their properties (conductivity and resistance) is much easier. It's a simple process. Depending on our requirements (dimensions of pellet) the amount of powder required can be calculated using simple calculations. Disc shaped pellets were prepared to study conductivity and resistance. To prepare these types of pellets we need to consider diameter, thickness and density for calculations.

Calculations:

Ohm's Law: V = I * R

V = Voltage; I = Current

Resistance (R):

$$R = \frac{\rho * T}{A}$$

 ρ = Resistivity ($\Omega * cm$); $A = Area (cm^2)$; T = Thickness (cm)

Conductivity (C):

$$C = \frac{1}{\rho}$$

Resistance is a property of a substance to resist the flow of current in an electrical circuit. Appropriate size was chosen and cleaned properly with acetone or

ethanol. Substrate used has a cylindrical hole in the middle which can be closed at one end and after placing powder into it, it can be closed on the other end also. Then mechanical force is applied (we have applied ~2 tones) to get it into disc shape. Graphene and its composites have good binding ability. But to have a real time evaluation these powders were mixed with polyester binder and Lead oxide powder as a pellet. These materials are used in the electrode preparation of lead acid batteries.





Figure 5.5 In-House development of electrical measurement of the test setup with Sample Preparation and Cell Assembly

5.5.1 Resistance calculation

Resistance is calculated using 2 probe technique. First of all, device is configured, and function is set with Electro-Chemical analyzer (Make: Wonatech, Southkorea). Sample is connected between 2 probes using 2 clips as shown in fig. 5.5. Measurement scans were started, and 04 different values are taken, average of which is taken as the resistance of the material and tabulated below (Table 5.3).

S.no	Method	Material Resistance (k Ω)
1.	Hummers	0.08
2.	Modified Hummers	0.02
3.	Microwave	0.0025

Table 5.3 Resistance Measurement

5.6 Surface Area Measurement

The surface area measurements were carried out with Coulter (3100), USA equipment. Where the Nitrogen (N₂) adsorption technique is followed to measure the surface area of the materials. The prepared graphene material has ~10 times of its original surface area. Table 5.4 discloses the increase in the surface area of the graphite material.

Table 5.4 Surface area measurement using N₂ adsorption.

S. No	Material	Surface Area (m ² /g)
1	Graphite	11.91
2	Expanded Graphite	16.47
5	Graphene - Microwave	128.99

5.7 Summary

In this chapter, various characterization tools have been used to validate the prepared graphene material. From the microscopic techniques it revealed that few layered graphene is observed. From the XRD spectra we could see the structural changes of graphite, expanded graphite and graphene in the step wise process. By analyzing the sample with Raman spectroscopy, the purity of the microwave is slightly higher with obtained material yield. The resistance of the graphene material is very much lower with the raw material (Graphite). Even the surface area has increased 10 times when compared to that of graphite. Therefore, we can corelate that surface area with the electrical conductivity, which is directly proportional. Now with successful preparation of the graphene material, it is taken forward to prepare the electrode as an additive to the negative active material to address the progressive failure (PGF). And also the same material is used to prepare a colloid suspension for coating the lead grids to address the Premature Failure (PMF) hindering the lead acid battery performances.

<u>Chapter – 6</u> Experimental Process/Investigations

As we discussed in earlier chapters, in 1860 by Gaston Plante first developed the first practical design of lead-acid battery, since then the production continued to grow. The majority usage of lead acid battery is by Automotive sector then followed by industrial sector like for traction and stand by application. Batteries are the primary customers for the lead producers around the world.

With the low cost and ease of manufacturing ability the lead-acid batteries demand is continuously growing when compared to other electrochemical devices.

The lead-acid battery having a Negative and positive electrode with spongy lead (Pb) and lead dioxide (PbO₂) is prepared with an electrolyte. On discharge, both electrodes convert to lead sulphate. Sulfuric acid solution has been used an electrolyte with the specific gravity of 1.29 (39% by weight). Below sections describes in brief about the components & its process of making a lead acid battery.

6.1 Manufacturing Process: Lead Acid Battery

The below flow chart (figure 6.1) describes the manufacturing process of the lead acid battery. We have followed jar formation in making the device, which involves in less defects, more feasible and ease of machining. These processes are discussed in detailed in below sections.

Device assembly is carried with several components as discussed in Chapter 1. But the real challenge would be preparing coated lead grids and formulating the active material with graphene. This process is called Plate preparation process (PLP). Followed with stacking with separator into the individual cells of 6 in a plastic container, this is termed as called components assembly process. Later these batteries are sent to acid filling process with desired specific gravity and sent to formation. Once this formation is completed, these are sent to finishing. With ensuring the electrical parameters the devices are ready for installation.



Fig. 6.1. Lead-acid battery manufacturing process.

6.1.1 Electrode Preparation

Preparing Positive & Negative Electrode consists of lead alloy as current collector and paste made up of lead oxide and additives vary for positive & negative.

One of the key roles of the battery performance are depends upon its electrode that is battery plate. The good quality battery plate with consistent quality in every lot having much and more demands in the market selection. of blade alloy or casting upgrade within lead acid batteries the importance of positive electrode controls performance songs battery failures is generally caused by degradation of positive plate hey pause 2 grids in all batteries act as both support structure and electrical conductor they must be capable of actions during production pause 2 grades are made from to hold the paste.

6.1.1.1 Lead Grid Preparation:



Fig. 6.2 Lead Ingots for grid fabrication

Through Gravity casting the lead grids are manufactured. Firstly, melting the lead ingots and pouring into the dye of our patented design as shown in below flow.



Gravity casting is the oldest process adopted by battery industry and invented in the year 1931 by Arthur D. Lund. It involves the pouring of molten alloy into mold with a gravitational force as shown in Fig. 6.3.



Fig. 6.3 Gravity casted regular lead grids.

This approach has few advantages, which are listed below.

- Cost effective
- Simple manufacturing process
- Quick solidification
- Uniform grid cooling

Our goal is to address the PMF and increase the battery performance, it is very much needed to prepare graphene solution to coat the lead grids. As the carbon cannot be coated with electro deposition technique, we choose to immersion/dip coating.

Chemical quantities may be used in preparing the graphene colloidal solution:

- 1. Demineralized (DM) Water 1000ml
- 2. Graphene 5gms
- 3. Sodium Cholate salt 5gms
- 4. Emeraldine salt 5gms

Procedure may be followed in preparing the colloidal solution:

 Graphene material is sonicated with Sodium cholate and Emeraldine salt for 60mins at ambient temperature.

- 2. Later all the materials are ball milled at 500RPM for 30mins in a planetary mill.
- Add all the milled particulates in 1000ml of Demineralized water and stir for 30mins to achieve colloidal solution.



Fig 6.4 Coating Process – (a) Dip coating/Immersion process (b) Graphene Solution.

Three step procedure has been has arrived and validated the coating effectiveness on the grids.

Step 1: Heat treatment of the grids @150°C for 30 Mins.

Step 2: Dip the grids in graphene Solution for 5Seconds.

Step 3: Drying the coated grids @105°C for 2Hours.



Fig. 6.5 Regular (Left) & coated grids (Right).

In detail Elemental analysis were carried out on coating importance and found out that the graphene participating

6.1.2 Active material pasting / Electrode preparation and Assembly

Along with the graphene, commercial-grade Binder, Blanc Fixe (BaSO₄) and Lignin is milled in a planetary ball mill at 500RPM for 15 Mins. This makes the material distribute and spread uniformly in the pasting process, The negative active material (PbO) was prepared at a temperature <65°C by mixing grey oxide with priorly prepared milled graphene materials, limiting the graphene to 0.2% of active material loading. This loading restriction has been based on the utilization of active material at formation. DM water was then added and made as a homogeneous mixture with a sulphuric acid solution of 1.40 specific gravity. The control batteries with commercial-grade carbon black of 0.2% are also prepared similarly, as mentioned above. After paste preparation, the lead alloy grids were machine-pasted with prepared pastes using a stainless-steel flat. The positive and negative electrodes were separated with AGM (Absorbed Glass Mat) separator, as shown in Fig. 3. Later, the electrical characterizations of battery has been carried out by multiple steps programmed galvanostatic charging with the help of DTN-01 (Bitrode Instruments, USA) with built-in voltage and temperature measuring device as per Japanese Industrial Standard (JIS-5302:2004).

Initial Reactions:

PbO (Active Material) +
$$H_2SO_4 = PbSO_4 + H_2O$$
 (1)

At curing the plate undergoes the following reaction:

 $(<70^{\circ}\text{C}) \text{ PbSO}_4 + 3\text{PbO} = 3\text{PbO} \cdot \text{PbSO}_4 (3\text{BS})$



Fig 6.6 Flow chart of Negative active material preparation.

(2)



Fig. 6.7 Electrodes (+Ve & -Ve) & Test Assembly of the cell components

6.1.3 Acid Filling

Sulphuric acid is used as an electrolyte in lead acid batteries. H₂SO₄ provides electrical conductivity for the electrochemical reactions to take place. It acts as ionic conductor (Hydrogen and sulphates ions) and provides a medium for transfer of electrons as ions inside the cell. Compatibility of the electrolyte with active materials is an essential factor in choosing the electrolyte. For example, aqueous solutions cannot be used in lithium cells and organic liquids cannot be employed in lead acid devices. The specific gravity of the electrolyte is 1.29, the volume of it is about 9 to 10 ml per Ah for other manufacturers they do have 11 to 15 ml per Ah along with other additives like phosphoric acid, for effective recombination's. The acid filling process is a automated process where a filling tube is inserted to the individual cell holes of the battery as shown in the below fig. 6.8.



Fig. 6.8 Acid Filling process

6.1.4 Formation & Finishing

Cured plates are formed in dilute sulphuric acid that is electrochemically converted to the respective active materials namely lead dioxide in positive plates and spongy Lead in negative electrodes. This is carried out in a dilute solution of sulphuric acid usually of 1.29 or 1.3 specific gravity made for made from pure battery grade acid. The plates must be fully submerged in the acid before the formation is started, the plates of opposite polarity must be separated with a separator with each other having enough uniform space between them so as to avoid direct contact with each other so that the electrodes here absorb the electrolyte. The formation is carried out with constant current of 0.1 to 1.0 amps over a period of 24 to 36 hours depending upon the capacity of the designed battery. Below reactions takes place at the time of formation. Lawrence et al., describes the effects of different soaking times of the battery at the formation stages best compositions are affected by the time between the formation and the soaking of the batteries. However electrical results and examination of the former best suggest that soaking time is to affect the resulting battery performance as shown in fig. 6.9.

Formation: At individual electrochemical reactions,

(+Ve electrode)
$$PbSO_4 + 2H_2O = PbO_2 + H_2SO_4 + 2e^- + 2H^+$$
 (a)

(-Ve electrode) $PbSO_4 + 2e^- + 2H^+ = Pb + H_2SO_4$ (b)



Fig. 6.9 Formation by soaking the batteries in water $<30^{\circ}C$

Below are the reactions happening in the batteries while performing in the field.

Discharging/Charging reactions:

-Ve electrode (Half Cycle):

+Ve electrode (Half Cycle):

$$PbO_2 + H_2SO_4 + 2H^+ + 2e^- \leftrightarrow PbSO_4 + 2H_2O (E^0 = 1.685 \text{ eV})$$
 (2)

Total voltage/potential across the electrodes: 0.356 + 1.685 = 2.041 eV

6.2 Process Validation

The process validation of the battery's will be carried out parallelly at the time of the component fabrications each step will be check it thoroughly and based on the specifications determine the quality of the components allowed for making the battery. In electrode (pasting process) preparation the battery electrodes are tested with the density of the active material and the vibration analysis of the plates. The active material, conversion of lead oxide powder to lead sulphates and also evaluating the free lead content in the active material is evaluated after the curing procedure.

In assembly, the main quality checks will be a round of the polarity check of the electrodes and then the intercell welding of the electrodes where all the electrodes are connected in series in the cells.

In acid filling, the lower limit and upper limit of the electrolyte is measured thoroughly before sending it for the formation if there is any low electrolyte feeling then the rest will be top-up, if there is an excess amount of electrolyte presence then those electrolytes will be dumped out of the design levels of the battery.

After formation, the electrodes are check for the conversion rate of the sulphates and lead oxide powder to lead dioxide in the positive plate and spongy lead in the negative plate. The best conversion rate will be considered for the manufacturing of the battery and also the High-Rate Discharge (HRD) test is carried out for the batteries to meet the requirement for dispatching the customers.

From our work to address premature failure, we validated the lead grids with and without coating process by the optical micrographs and elemental analysis by EDAX.

6.3 PMF: Optical Micrograph

In Fig. 10 an optical micrograph (Nikon Optima, Japan) reveals us the coating thickness with the black line on the surface of the lead grid with ~20 microns.



Fig. 6.10 Optical Micrographs of the lead grid surface.

(a) Uncoated grid (b) Graphene coated grid with thickness ~ 20microns.

6.4 PMF: Energy Dispersive X-Ray Analysis (EDAX)

Elemental analysis of the uncoated and coated lead grids is captured (Fig. 6.11). Further the grids were pasted and cured as per the process. These cured plates are later verified for the formation of sulphates on the grids.

In prior art by M. Saravanan et. al., (2015) provided in Fig. 2.11, there the presence of sulphur after coating was observed, which creates the raise of resistance in the electrode. But, whereas with our approach it is evident that with the successful coating the grids have effectively restricted the formation of sulphur/sulphates on the grids (Fig. 6.11(b) & 6.12(b)) and expected a better battery performance.





Fig. 6.11 Elemental analysis of the grid (a) Regular / Uncoated Grid (b) Graphene coated grid.





Fig. 6.12 Elemental analysis of the cured electrode (a) Regular / Uncoated Grid (b) Graphene coated grid.

For addressing the progressive failure (PGF) we prepared batteries with graphene aided negative active material and the electrodes are evaluated by Cyclic voltammetry and Impedance analysis.

6.5 PGF: Cyclic Voltammetry

The prepared electrodes are subjected to cyclic voltammetry studies with the equipment supplied by Wonatech, Korea. The counter electrode used was a platinum electrode and the reference electrode was Standard Calomel Electrode (SCE). The sweep potentials were maintained from -1.0V to 2.5V with a scan rate of 50mV of the active area 1cm². For Lead Alloy analysis: Platinum was used as a counter electrode. For Electrode Plates: +Ve/-Ve plate of the lead acid cell was used as the counter electrode. Keeping calomel electrode as a reference electrode for all the electrochemical analysis. The initial rise of the current peak ~0.2A at -0.5V

show in the Fig. 6.13 is the evidence that the prepared electrode is porous tend the electrolyte accessibility for reversible redox reaction, when compared to the control electrodes. It is also understood that the free lead present is converted to PbSO₄ by the addition graphene. Based on this the reactivity of the Graphene electrodes are better than the control one. The measured specific capacitance from the CV curve with the below equation (1) of the Graphene electrode is ~0.63 F/gm while carbon black-based control electrodes are of ~0.52F/gm, where a ~21% higher specific capacitance is observed with Gr electrode.

Calculating The Specific Capacitance by voltammetric study from below equation 1.

 $C = c/m (c=Q/V; Q=\int A/v) - (1)$

- c = capacitance
- m = Mass (g)
- Q = charge
- A = Area
- v = Scan rate
- V = Potential Window



Fig 6.13 CV curves of Control and Graphene (Gr) electrode with experimental setup to the right

6.6 PGF: Impedance Spectroscopy

EIS are performed with the equipment Zive SP1, South Korea and Nyquist plots are developed by two electrode mode with graphene and without graphene. The EIS test was carried out with the amplitude of 10mV and the frequency ranging 1KHz to 0.1Hz. First, thing noticed from the plot that the graphene curves are much lower when compared to the control plot. This highlights the resistance of the gr electrode to be low to that of control. This means that the conductivity of the Gr electrode is high and dispersed uniformly. The illustration from the Fig. 6.14 gives us the parameters associated to the electrode behaviour. The electrode electrolyte interference resistance, Electrode charge transfer resistance and Warburg resistance and obtained and tabulated in Table 1. These results suggest that the addition of Graphene significantly reduces resistance developed from the reaction w.r.t to the electrolyte and electrode surface promoting the redox reaction more effectively. The Warburg Resistance is observed lower attributing to the porous structure, which promotes the ions diffusivity. Lower W leverage higher diffusivity between electrode and electrolyte interface.



Fig 6.14 Nyquist Plot of control and graphene electrode with the experimental setup to the right (Inset: Equivalent Circuit).

Parameters	Control	Graphene (Gr)
R _s (Electrode/Electrolyte	11.64	2.97
Resistance) Ω		
R _{ct} (Charge Transfer	99.32	59.64
Resistance) Ω		
W (Warburg Resistance) Ω	0.04	0.01
Q (Capacitance) F	11.38	18.94

Table 1. EIS parameters of control and graphene electrode.

6.7 Summary

Electrodes with carbon coated grids are prepared and analysed before assembling into the device. Similarly, the electrodes with graphene aided negative active material were also prepared and evaluated before going for battery assembly. The analysis has confirmed the desired electrodes are in better position than the regular and can take for assembling the batteries. The coated batteries for stationary application are prepared and the batteries with graphene in negative active material were prepared for automotive application for further electrical assessment. With both approaches the batteries are made with AGM technology. The advantage of an AGM technology battery search that the batteries do not require water addition throughout their life & also user-friendly battery in the sense that evolves negligible number of gases in the normal course of operation and there is no acid seepage.

<u>Chapter – 7</u> <u>Results & Discussion</u>

There are several standards to be complied when an energy storage device is made, to ensure the prepared device is ready to meet the specific requirements like discharge capacity, charge acceptance, cranking ability and life cycling w.r.t application. These standards include International Electrotechnical Commission Standard (IEC); the British Standard (BS/EN); Deutsches Institut für Normung Standard (DIN); The Japanese Industrial Standard (JIS); Telecommunication Engineering Commission (TEC), and so forth. As IEC/TEC & JIS are widely known standards for the stationary and Automotive applications and hence were followed for our studies.

7.1 Application of Graphene to improve Lead Acid Battery Performance

For Stationary Application (PMF):

The validations of high current discharges include 10, 8, 5, 1 hour, 15 Minutes and 5 Minutes tests on the 2V lead acid cells with graphene coated and uncoated lead grids have been conducted.

For Automotive Application (PGF):

1) Capacity/Energy density: The capacity of automotive batteries is rated at 10 hours, which means that the battery should last for the stated duration when discharged at a given current. For instance, a battery with a capacity of 3 Ah should last for 10 hours as it discharges at a current of 0.3A.

2) High Rate Discharge (HRD) Test: To evaluate the cranking ability of the battery at cold condition (-15°C).

3) Charge Acceptance Test (CAT): To estimate the time taken for recharging the battery.

The rating tests were carried out at Amararaja Batteries Limited, Tirupathi, India. Table 7.1 shows the characteristics of battery samples being prepared with positive coated grids.

7.1.1 PMF: Electrical Performance of positive Coated grids (2V-660Ah)

The capacity test of the cells with high current discharge were carried out. The C10, C8 and C5 hr test showed that the cells with coat and uncoated produced more than ~100% capacity with ease. But at C1 hr test the regular cells were found to be struggling and produced marginal higher to 100% capacity. Whereas graphene coated grids have comfortably delivered ~115%. When comes to C15 mins the regular cells were able to deliver only ~75% capacity and the coated batteries were able to deliver ~154% capacity, which is significant superior. And when test for C5 mins capability the regular grids were just limited to ~56% and the graphene coated cells continued to deliver ~156% capacity. Overall, the cells with graphene coat on the positive grids have performed better than the regular uncoated grids as presented in below graph Fig. 7.1. This significant improvement could be very much useful for peak load requirement at on/off grid stationary application.



Fig. 7.1 Electrical assessments of coated and uncoated lead grids with 03 times

repetition.

7.1.2 PGF: Electrical Performance of Graphene in NAM (12V-3Ah)

(a) Discharge Capacity:

The performance of the battery was tested for its discharge capacity in the range of Cn/10hr. The current test rating is expressed in terms of the nominal capacity of the battery, calculated at 95% utilization of the active material. Fig. 7.3 shows the comparison between the discharge capacity of batteries comprising reduced graphene oxide with that of control batteries. All the tests were conducted at 25±20C. These batteries were discharged with 0.1C rate to a cut-off voltage of 10.5V. It is observed that the discharge time of test batteries with graphene oxide is longer compared to that of the control batteries. Further, it is observed that the current discharge capacity increased by nearly 27% when Graphene is added to NAM in comparison to control batteries. These improvements are attributed to the conversion of the active material to a large extent.

Test Conditions:

Environment: $25^{\circ}C \pm 2^{\circ}C$; Discharge with I10 Current; Cutoff voltage 10.5V; Duration t(h) recorded C10 (Ah) = I10 * t



Figure 7.3: Discharge Capacity of Control vs Graphene battery.

Observations:

- 1. Uniform conductive network,
- 2. Higher active material utilization, due to small & porous structure.
- 3. Maintains the thirst for active mass towards acid.
- 4. Improved discharge capacity (~27%).

(b) Charge Acceptance (CAT):

Charge acceptance is one of the important performance characteristics for batteries in applications involving high depth of discharge (DoD). A higher charge acceptance capacity of a battery extends battery cycling life. The charge acceptance is measured at the 10th minute by applying a constant voltage of 14.4V. The observations from fig 7.4., confirms an enhancement of 14% in charge acceptance is observed when compared to control batteries by adding Graphene to NAM. Initially, the lower current intake was observed with the Graphene battery, this is due to the resistance buildup in the electrodes. With the intake of the current, the sulfates causing the resistance is reduced over the period and later increase the current intake. This is due to the availability of adequate Graphene with a unique flexible porous structure that promotes better transport of hydrogen and sulfate ions. Further, it promotes improved utilization of active mass and uniform distribution of sulfates.

Test Conditions:

Environment: $25^{\circ}C \pm 2^{\circ}C$; Discharge with I10 Current, for 5hrs; Store at $0^{\circ}C \pm 2^{\circ}C$ for 6hrs; Charge with 14.4V ± 0.1V @ 10min Duration is recorded.



Figure 7.4: Charge Acceptance of Control vs Graphene battery

Observation:

Unique porous structure, Facilitates H+ & SO4- ions, Uniform distribution of sulphates & Improves active material utilization. Improved charge acceptance (~10%).

(c) Cranking Ability (HRD):

High-rate current discharge attributes to the cranking ability when a battery is discharged at comparatively high current as per the storage capacity of battery. HRD feature is carried out as per the conditions, to measure the voltage, after the start of discharge and till the completion time. Generally, the batteries meet the 5thsec voltage requirement as per JIS standard; hence, we have not mentioned. But the time taken to reach 6V, while discharging @ -15°C fumbles, this has been addressed in the following. Battery (12V/3Ah) life gets affected when the discharge current exceeds the rated discharge current during the vehicle start-up. Hence the study on high-current (30A) discharge assumes importance in characterizing battery performance. It is observed from fig 7.5. That the time taken for the discharge capacity of the batteries with Graphene exceeds the requirement and for a long-time-interval of 23% (23s). This represents the performance improvement over control batteries in terms of high-rate current discharge.

Test Conditions:

Environment: $-15^{\circ}C \pm 2^{\circ}C$ for 16hrs; Discharge with 30A; Cutoff voltage 6V; Duration t(min) recorded.



Figure 7.5: Cold Cranking ability of Control vs Graphene battery

Observation:

At HRD, discharge current exceeds rated discharge, which impacts capacity. Reaction happens only on the surface of the electrode, which lead to large sulfates formation, hindering the conductivity. Improved cranking ability (~23%).

(d) Life Cycle Performance:

The cycle life of a battery (12V/3Ah) is defined as the complete number charging (0.6A) and complete discharging (3A) cycles, before its capacity drops to ~30% of the original capacity. This test is concerned with the frequency (number of times) at a repetition of charging and discharging of high load area with discharge depth of 100% of each cycle and limiting to 30% State of Health (SoH) cut-off condition, making this as one of the critical battery performance indicators.

Test Conditions:

Environment: 40°C ± 2°C; Discharge: Current 3A, for 1hr, Cutoff voltage 10.2V; Charge: Current 0.6A, for 5hrs, Repeat for 25 cycles then Run Capacity Test @40%DoD, after every 25 cycles; Duration is recorded. Run the test until <50% SoH of 80% rated capacity.

86



Figure 7.6: Life cycling ability Control battery (Left) and Graphene Battery (right)

Fig. 7.6 shows the cycling behavior of Gr batteries at a 1C discharge rate. It is observed that at 225 cycles, the control batteries show a drop in the discharge capacity and failed to recover. Whereas the cells with Graphene layers in the negative active material performed up to 300 Cycles with a stable discharge. This may be attributed to lowering the internal resistance of the cells due to limiting the formation of larger sulfate crystals. From this investigation, it is observed that by adding, Gr in NAM increased charge acceptance as well as restricted the sulfation in the negative electrodes during the deep discharge cycling of the batteries. The charge /discharge cycling of the batteries, with Graphene layers, have a ~33% enhancement when compared to that of the control batteries. From Fig. 7.7, SEM images of the negative electrode with Gr have relatively small PbSO₄ when compared with Control electrodes. We can conclude that with the addition of Gr in NAM, we can increase the active material utilization with enhanced discharge capacity and higher life cycles. Higher life of the battery gives us an additional benefit of cost for replacing the battery after its end of life. Adding an advantage to the environment towards recycling.





Fig 7.7. SEM Images after end cycles a) Control b) Gr

7.2 Summary

In the present investigation, graphene is prepared using exfoliated graphite through the microwave irradiation technique. Performance studies are conducted on Lead Acid Battery to determine the effect of graphene as a coating material for stationary application to successfully restrict the premature failure (PMF). Othe area, the progressive failure (PGF) has been addressed by using graphene as an additive in Negative Active Material (NAM) for automotive (2Wheeler) application. The performance test results are compared with the performance of control batteries. The effects of the addition of this graphene material on battery performance like discharge capacity, charge acceptance, cranking ability & cycling performance were investigated and found better.

Premature Failure (PMF):

The 2V-600Ah lead acid cells developed with graphene coated grids have enhanced the high current discharge at short durations of C15 Mins – 2x times and C5 Mins – 3x times when compared to the control cells. This significant improvement was achieved by limiting the growth of the sulfate on the girds and providing a conductive network at high current discharges.

Progressive Failure (PGF):

When graphene is used as an additive in the negative active material and developed a 12V-3Ah batteries, the discharge capacity has enhanced (~27%) to a large extent with a cranking ability higher of ~23%. In addition to this, the batteries with the graphene exhibit ~33% higher stable life cycle performance. This significant improvement was observed by limiting the irreversible sulfation of the negative electrode, improving its charge acceptance ability (~14%) by providing a conductive network during charge and discharge cycles.

The emphasis on graphene as an additive material for lead acid batteries is our first attempt, and this could lay a platform to develop advanced lead-acid batteries for high-rate partial state of charge applications like HEV/EV's.

<u>CHAPTER - 8</u> <u>CONCLUSIONS</u>

In this chapter, salient features of our work have been highlighted, and an attempt to summarize our research has been made. In addition to this, a broad vision for future research work has also been provided.

8.1 Summary of the present work

This chapter presents the overall conclusions from experimental investigations conducted on Absorbed Glass Mat Lead Acid Energy Storages are presented. In this work, our scope is to improve the electrical performance of the lead-acid energy storages with graphene material by addressing two critical failures happening in the battery (1) Premature Failure and (2) Progressive Failure for stationary and automotive applications respectively.

In the present investigation, graphene material is studied to address the premature and progressive failure of energy storage. In this thesis, preparation of Graphene material, characterization and analysis of graphene-based electrodes subjected to the electrical characteristics were carried out. In the investigation process, two approaches are investigated (1) Colloidal form of graphene for premature failure & (2) Powder form of Graphene for Progressive failure of the energy storages.

Taking forward with Electrodes: Preparation, characterization, assembly and electrical performances has been done and the results are presented. The electrical performances like Initial Capacity (Energy Density), Charge Acceptance (Recharging behavior), Cold Cranking (High current discharges), and Life cycling test were performed on the graphene-based devices and have been compared with the devices without graphene material. The essential conclusions drawn from this present work are as follows.

The preparation of materials was crucial since it can affect the electrode characteristics considerably. The study concluded that the 2-dimensional material is found that its contribution towards the electrode's performance has enhanced significantly in a cost-effective way.

Various methods for synthesizing graphene were analyzed and optimized for a novel approach to synthesizing graphene in bulk with cost-effectiveness. Using the aqua regime, graphite was partially oxidized and then exposed to microwave radiation to get graphene worms. By sonicating these worms in the solvent environment until a uniform solution is achieved, followed by solvent evaporation to collect our desired 2d material.

In conclusion, initially using the Hummers method Graphite Oxide (GO) was prepared. Further exposure of GO to microwave irradiation followed by ultrasonication we obtained with Few Layered Graphene (FLG). The synthesized material's cross-sectional morphology, structure, and phase studies were carried out using many characterization tools like Surface area analyzer, scanning electron microscopy (SEM), x-ray diffraction, transmission electron microscopy, and Raman spectroscopy. Resistance values of the prepared material were calculated with an in-house developed test rig setup.

Due to its outstanding physical and chemical properties, graphene plays a good role as an electrode material and makes it possible to produce highperformance energy storage devices (Lead Acid).

91
The essential conclusions drawn from the present work are as follows:

(a) Premature Failure (PMF)

Positive grid corrosion is the phenomenon revealed that the cause of the failure is the formation of a barrier layer of lead sulfate growth between the grid and the active material. To restrict the formation of the passivation layer, we have coated the grids with a graphene-based solution and found that the coated grids are more corrosive resistant than the uncoated grid. The grid growth studies were extensively carried out.

(b) Progressive Failure (PGF)

The optimal composition of graphene with an active material to achieve the device's maximum efficiency was critical. Several compositions were analyzed in terms of Elemental concentration, conductivity, and pH. After identifying the optimal composition and preparing the electrodes, quality checks were carried out. After analyzing the electrodes in terms of active material conversions, these electrodes were further studied with Cyclic voltammetry and Electrochemical impedance spectroscopy. After the evaluation, the electrodes were assembled connected in series to form a 12V battery for further electrical evaluation.

Keywords: Carbon; Nanomaterial; Graphene; Lead Acid Battery; Lead Grids; Negative Active Material; Premature Failure (PMF); Progressive Failure (PGF); Sulphation; Stationary Lead Acid Battery; Automotive Lead Acid Battery; Japanese Industrial Standard (JIS); High Current discharges; Energy density (Capacity); Charge Acceptance; Cranking Ability; Charge/Discharge Cycles.

92

The outcomes of this experimental investigations from the synthesis of different nanomaterials to applications are listed below. And found in the below link.

https://scholar.google.co.in/citations?hl=en&user=OD8c83cAAAAJ&view_op=list_w orks&sortby=pubdate

Publications:

- Effect of Milled Carbon as Negative Electrode Additive for Lead Acid Energy Storage Device – 2020.
- 2. Synthesis and study of reduced graphene oxide layers under microwave irradiation 2018.
- Facile synthesis and thermal analysis of antimony telluride nanostructures 2018.
- Synthesis of dendritic-flowers of wurtzite Cu₂ZnSnS₄ via solvothermal process - 2017.
- Synthesis of Graphene Oxide by Modified Hummers Method and Hydrothermal Synthesis of Graphene-NiO Nano Composite for Supercapacitor Application – 2016.
- Synthesis and analysis of ternary oxide nanomaterial for electro-thermal applications - 2015.
- Synthesis and evaluation of zinc-based nanomaterials for optical-thermal studies – 2015.

8.2 Scope for future work

With the success of this work in identifying potential areas of improving the electrochemical behavior attributed to the grid (Current collector) surface and with unique morphology of low dimensional material used along with the lead oxide as a precursor for lead-acid batteries has revealed the future opportunities and efforts to fulfill the global objective and needs in producing a prototype of lead-acid hybrid supercapacitor battery.

Based on the results from this work, future research will emphasize the following:

- The carbon-coated lead grids with different coating thickness for HEV & EV applications.
- The mesoporous carbon materials are to be extended into the making of producing electrodes to meet objective cost and scalable production feasibility.
- Identifying a common reducing material of the sulfation formed on electrode surface due to the interaction with the electrolyte.
- Constructing a prototype of lead-acid hybrid battery with supercapacitor behavior with the carbon materials in both positive & negative electrodes for evaluation.
- Develop a battery with No lead grids by replacing with Carbon electrode.

REFERENCES

- A. A. Balandin, S. Ghosh, W. Bao, I. Calizo, D. Teweldebrhan, F. Miao, and C. N. Lau "Superior Thermal Conductivity of Single-Layer Graphene", Nano Letters, 2008, 8, 902.
- 2. A. Dato, V. Radmilovic, Z. Lee, J. Philips, and M. Frenklach "Substrate-Free Gas-Phase Synthesis of Graphene Sheets", Nano Letters, 2008, 8, 2012.
- A. Ghosh, Y.H. Lee, Carbon-Based Electrochemical Capacitors, ChemSusChem. 5 (2012) 480–499. doi:10.1002/cssc.201100645.
- 4. A. H. Castro Neto, F. Guinea, and K. S. Novoselov and A. K. Geim "The electronic properties of graphene" REVIEWS OF MODERN PHYSICS, VOLUME 81, JANUARY-MARCH 2009.
- A. Rudge, J. Davey, I. Raistrick, S. Gottesfeld, J.P. Ferraris, Conducting polymers as active materials in electrochemical capacitors, Journal of Power Sources. 47 (1994) 89–107. doi:10.1016/0378-7753(94)80053-7.A.J. Salkind "A historical overview of battery Innovation and Technology compared with Ancillary Science" in Pro. Symp. History of battery Technology, ed A.J. Salkind, Proc, Vol87-14, The Electrochemical Society., N.J, USA, 1987,p.1.
- A.F. Hollenkamp, "Secondary batteries –lead-acid systems: carbon additives", Encyclopedia of Electrochemical Power Sources, Elsevier Science, 2009, pp. 638-647 (doi:10.1016/B978-044452745-5.00152-0).
- A. Kaniyoor, T.T. Baby, S. Ramaprabhu "Graphene synthesis via hydrogen induced low temperature exfoliation of graphitic oxide" Journal of Materials Chemistry 20 [8467-8469] (2010).
- A. K. Geim and K. S. Novoselov "The Rise of Graphene", Nature Materials, 2007, 6 (3), 183.
- 9. Arrla Eswaraiah, Sasidharannair Sasikaladevi Jyothirmayee Aravind and Sundara Ramaprabhu * arrla Eswaraiah, Sasidharannair Sasikaladevi Jyothirmayee Aravind and Sundara Ramaprabhu * "Top down method for Synthesis of Highly Conducting Graphene by Exfoliation of Graphite Oxide using Focused Solar Radiation" Journal of Materials Chemistry, The Royal Society of Chemistry 2011.
- 10.A. Reina, X. Jia, J. Ho, D. Nezich, H. Son, V. Bulovic, M. S. Dresselhaus, and J. Kong "Large Area, Few-Layer Graphene Films on Arbitrary Substrates by Chemical Vapor Deposition" Nano Letters, 2009, 9, 30.
- 11.A. Thess, R. Lee, P. Nikolaev, H. Dai, P. Petit, J. Robert, et al., Crystalline Ropes of Metallic Carbon Nanotubes, Science. 273 (1996) 483–487. doi:10.1126/science.273.5274.483.
- 12.Bin Shen, Wentao Zhai,* Dingding Lu, Wenge Zheng* and Qing Yan "Fabrication of microcellular polymer/graphene nanocompositefoams" June 2012, DOI 10.1002/pi.4355.
- 13.B. C. Brodie. "On the Atomic Weight of Graphite", Proceedings of the Royal Society of London, 1859, 10, 249.
- 14.B.C. Thompson, J.M.J. Fréchet, Polymer–Fullerene Composite Solar Cells, Angewandte Chemie International Edition. 47 (2008) 58–77. doi:10.1002/anie.200702506.

- 15.B.J. Landi, M.J. Ganter, C.M. Schauerman, C.D. Cress, R.P. Raffaelle, Lithium Ion Capacity of Single Wall Carbon Nanotube Paper Electrodes, J. Phys. Chem. C. 112 (2008) 7509–7515. doi:10.1021/jp710921k.
- 16.B.J. Landi, M.J. Ganter, C.D. Cress, R.A. DiLeo, R.P. Raffaelle, Carbon nanotubes for lithium ion batteries, Energy Environ. Sci. 2 (2009) 638–654. doi:10.1039/B904116H.
- 17.B.-J. Yoon, S.-H. Jeong, K.-H. Lee, H. Seok Kim, C. Gyung Park, J. Hun Han, Electrical properties of electrical double layer capacitors with integrated carbon nanotube electrodes, Chemical Physics Letters. 388 (2004) 170–174. doi:10.1016/j.cplett.2004.02.071.
- 18. Cardinal components-White Paper on "Design Advantages of Solid State Batteries versus Supercapacitors" CC-SSB-16 December 2013.
- 19.C. Liu, Z. Yu, D. Neff, A. Zhamu, B.Z. Jang, Graphene-Based Supercapacitor with an Ultrahigh Energy Density, Nano Lett. 10 (2010) 4863–4868. doi:10.1021/nl102661q.
- 20.C. Lee, X. Wei, J. W. Kysar, and J. Hone "Measurement of the Elastic Properties and Intrinsic Strength of Monolayer Graphene", Science, 2008, 321, 385.
- 21.Cooper Bussman (powersoft) a white paper on "High Power Supercapacitors Enable New Power Applications and Replace Batteries" May 2010.
- 22. Daniela C. Marcano, Dmitry V. Kosynkin, Jacob M. Berlin, Alexander Sinitskii, Zhengzong Sun, Alexander Slesarev, Lawrence B. Alemany, Wei Lu, and James M. Tour* "Improved Synthesis of Graphene Oxide" 10.1021/nn1006368 June 2010.
- 23.Dale A.C. Brownson, Dimitrios K. Kampouris, Craig E. Banks* "An overview of graphene in energy production and storage applications" Journal of Power Sources 196 (2011) 4873–4885.
- 24.David Allen, Chris Brown, John Hickey, Vinh Le, Robert Safuto "New York Independent System Operator "Energy Storage in the New York Electricity Market" New york independent system operator, draft for discussion-december 2009.
- 25.D.B. Le, S. Passerini, J. Guo, J. Ressler, B.B. Owens, W.H. Smyrl, High Surface Area V 2 O 5 Aerogel Intercalation Electrodes, J. Electrochem. Soc. 143 (1996) 2099– 2104. doi:10.1149/1.1836965.
- 26.D.C. Marcano et al "Improved synthesis of graphene oxide" Journal of the American Chemical Society 4 [4806-4814] (2010).
- 27.D.D.L Chung "Review of exfoliation of graphite", Journal of Materials Science 22 [4190-4198] (1987).
- 28.DG ENER Working Paper "The future role and challenges of Energy Storage" European commission directorate general for energy, DG ENER working paper April 2013.
- 29.D. Pavlov, P. Nikolov, and T. Rogachev, "Influence of carbons on the structure of the negative active material of lead-acid batteries and on battery performance", Journal of Power Sources, 196, 5155 (2011) (doi:10.1016/j.jpowsour.2011.02.014).
- 30.D. Povlov, P. Nikolov, "Capacitive carbon and electrochemical lead electrode systems at the negative plates of lead-acid batteries and elementary processes on cycling", J. Power Sources 242 (2013) 380–399 (doi: 10.1016/j.jpowsour.2013.05.065).
- 31.D. Puthusseri, V. Aravindan, S. Madhavi, S. Ogale, 3D micro-porous conducting carbon beehive by single step polymer carbonization for high performance

supercapacitors: the magic of in situ porogen formation, Energy Environ. Sci. 7 (2014) 728–735. doi:10.1039/C3EE42551G.

- 32.D.D.L Chung "Review of exfoliation of graphite", Journal of Materials Science 22 [4190-4198] (1987).
- 33.D. R. Dreyer, S. Park, C. W. Bielawski, and R. S. Ruoff "The Chemistry of Graphene Oxide", American Journal of Physical Chemistry, 2006, B110, 8535.
- 34.D.S. Hecht, L. Hu, G. Irvin, Emerging Transparent Electrodes Based on Thin Films ofCarbon Nanotubes, Graphene, and Metallic Nanostructures, Adv. Mater. 23 (2011) 1482–1513. doi:10.1002/adma.201003188.
- 35.Edward L. Wolf "Nanophysics and Nanotechnology: An Introduction to Modern Concepts in Nanoscience", chemical education today, 174 pp.ISBN 3527404074, 2004.
- 36.E. Frackowiak, F. Béguin, "Electrochemical storage of energy in carbon nanotubes and nanostructured carbons", Carbon 40 (2002) 1775-1787 (doi: 10.1016/S0008-6223(02)00045-3).
- 37.E. Frackowiak, F. Béguin, Carbon materials for the electrochemical storage of energy in capacitors, Carbon. 39 (2001) 937–950. doi:10.1016/S0008-6223(00)00183-4.
- 38.Jakub Lach, Kamil Wróbel, Justyna Wróbel, Piotr Podsadni & Andrzej Czerwiński "Applications of carbon in lead-acid batteries: a review" Journal of Solid State Electrochemistry volume 23, pages693–705, 2019. (doi.org/10.1007/s10008-018-04174-5).
- 39.Gedela Venkata Ramana, Mokhtar Ali, and Vadali V. S. S. Srikanth* "Carbon- and Polyaniline- Nanofibers Containing Composite Electrode Material for Supercapacitors" J. Nanosci. Nanotechnol.14, 1–6, 2014.
- 40.G. Venkata Ramana, Balaji Padya, Vadali V.S.S. Srikanth, P.K. Jain*, G. Padmanabham, and G. Sundararajan "Electrically conductive carbon nanopipe-graphite nanosheet/polyaniline composites" CARBON49 (2011) 5239–5245
- 41.H. Hu, X. Wang, J. Wang, L. Wan, F. Liu, H. Zheng, R. Chen, and C. Xu "Preparation and Properties of Graphene Nanosheets-Polystyrene Nanocomposites via in situ emulsion polymerization", Chemical Physics Letters, 2010, 484, 247.
- 42. Hofmann "Advanced nanomaterials" Powder Technology Laboratory, IMX, EPFL Version 1 Sept 2009.
- 43.Hong-Chang Tian, Jing-Quan Liu*, Dai-Xu Wei "Graphene oxide doped conducting polymer nanocomposite film for electrode-tissue interface" Biomaterials 35 (2014) 2120e2129.
- 44.HughO.Pierson "Handbook of carbon, graphite, diamond and fullereneproperties,processing and applications" Consultant and Sandia National Laboratories (retired) June 2003.
- 45.In Kyu moon, Junghyun Lee, Rodney s. Ruoff & Hyoyoung Lee "Reduced graphene oxide by chemical graphitization" nature communications DOI: 10.1038/ncomms1067.
- 46.Imre P. Gyuk U. S. Department of Energy (Washington D.C) "EPRI-DOE Handbook of Energy Storage for Transmission & Distribution Applications" 1001834 Final Report, December 2003.
- 47.I.-H. Kim, J.-H. Kim, B.-W. Cho, Y.-H. Lee, K.-B. Kim, Synthesis and Electrochemical Characterization of Vanadium Oxide on Carbon Nanotube Film Substrate for

Pseudocapacitor Applications, J. Electrochem. Soc. 153 (2006) A989–A996. doi:10.1149/1.2188307.

- 48.J. C-Delgado, J. M. R-Herrera, X. Jia, D. A. Cullen, H. muramatsu, Y, T., M. Endo, H. Terrones, M. S. Dresselhaus, and M. Terrones "Bulk Production of a New Form of sp2 Carbon: Crystalline Graphene Nanoribbons", Nano Letters, 2008, 8, 2773.
- 49.Jeffrey Jordana, Karl I. Jacobb, Rina Tannenbaumc, Mohammed A. Sharafb, Iwona Jasiukd,* "Experimental trends in polymer nanocomposites—a review", Materials Science and Engineering A 393 (2005) 1–11.
- 50.J.O. Besenhard, H.P. Fritz, Cathodic reduction of graphite in organic solutions of alkali and NR4+ salts, Journal of Electroanalytical Chemistry and Interfacial Electrochemistry. 53 (1974) 329–333. doi:10.1016/S0022-0728(74)80146-4.
- 51.J. Liang, Y. Wang, Y. Huang, Y. Ma, Z. Liu, J. Cai, C. Zhang, H. Gao, and Y. Chen "Electromagnetic Interference Shielding of Graphene-Epoxy Composites", Carbon, 2009, 47, 922.
- 52.J.P. Paraknowitsch, A. Thomas, Doping carbons beyond nitrogen: an overview of advanced heteroatom doped carbons with boron, sulphur and phosphorus for energy applications, Energy Environ. Sci. 6 (2013) 2839–2855. doi:10.1039/C3EE41444B.
- 53.J.P. Zheng, P.J. Cygan, T.R. Jow, Hydrous Ruthenium Oxide as an Electrode Material for Electrochemical Capacitors, J. Electrochem. Soc. 142 (1995) 2699–2703. doi:10.1149/1.2050077.
- 54.J.R. Dahn, T. Zheng, Y. Liu, J.S. Xue, Mechanisms for Lithium Insertion in Carbonaceous Materials, Science. 270 (1995) 590–593. doi:10.1126/science.270.5236.590.
- 55.Juan Zhang, Tian Tian, Youhu Chen, Yufang Niu, Jie Tang, Lu-Chang Qin* "Synthesis of graphene from dry ice in flames and its application in supercapacitors" Chemical Physics Letters 591 (2014) 78–81.
- 56.K. Fukuda, K. Kikuya, K. Isono, M. Yoshio, Foliated natural graphite as the anode material for rechargeable lithium-ion cells, Journal of Power Sources. 69 (1997) 165–168. doi:10.1016/S0378-7753(97)02568-8.
- 57.K. Hata, D.N. Futaba, K. Mizuno, T. Namai, M. Yumura, S. Iijima, Water-Assisted Highly Efficient Synthesis of Impurity-Free Single-Walled Carbon Nanotubes, Science. 306 (2004) 1362–1364. doi:10.1126/science.1104962.
- 58.K. Mizushima, P.C. Jones, P.J. Wiseman, J.B. Goodenough, LixCoO2 (0<x<-1): A new cathode material for batteries of high energy density, Materials Research Bulletin. 15 (1980) 783–789. doi:10.1016/0025-5408(80)90012-4.</p>
- 59.K. Nakamura, M. Shiomi, K. Takahashi, M. Tsubota, "Failure modes of valveregulated lead/acid batteries", J. Power Sources 59 (1996) 153-157 (doi: 10.1016/0378-7753(95)02317-8).
- 60.K. Sato, M. Noguchi, A. Demachi, N. Oki, M. Endo, A Mechanism of Lithium Storage in Disordered Carbons, Science. 264 (1994) 556–558. doi:10.1126/science.264.5158.556.
- 61.L. Dai, D.W. Chang, J.-B. Baek, W. Lu, Carbon Nanomaterials for Advanced Energy Conversion and Storage, Small. 8 (2012) 1130–1166. doi:10.1002/smll.201101594.
- 62.Lingyu Sun, Ronald F. Gibson,*, Faramarz Gordaninejad, and Jonghwan Suhr "Energy absorption capability of nanocomposites: A review" Composites Science and Technology 69 (2009) 2392–2409, doi:10.1016/j.compscitech.2009.06.020.

- 63.L.L. Zhang, Z. Lei, J. Zhang, X. Tian, X.S. Zhao, Supercapacitors: Electrode Materials Aspects, in: Encyclopedia of Inorganic Chemistry, JohnWiley&Sons,Ltd,2006. doi:10.1002/0470862106.ia816/abstract.
- 64.L.L. Zhang, R. Zhou, X.S. Zhao, Graphene-based materials as supercapacitor electrodes, J. Mater. Chem. 20 (2010) 5983–5992. doi:10.1039/C000417K.
- 65.LIMING DAI* "Functionalization of Graphene for Efficient Energy Conversion and Storage" accounts of chemical research, received on April 29, 2012.
- 66.Luke S. Walker, Victoria R. Marotto, Mohammad A. Rafiee, Nikhil Koratkar, and Erica L. Corral* "Toughening in Graphene Ceramic Composites", Published online 10.1021/nn200319d March 2011.
- 67.Manja Kurecic and Majda Sfiligoj Smole "Polymer Nanocomposite Hydrogels for Water Purification", http://dx.doi.org/10.5772/51055.
- 68.Marin S. Halper, James C. Ellenbogen "Supercapacitors: A Brief Overview", MP 05W000027 March 2006.
- 69. M Kintner-Meyer, M Elizondo, P Balducci, Viswanathan.V, C Jin, X Guo, T Nguyen, and F Tuffner "Energy Storage for Power Systems Applications: A Regional Assessment for the Northwest Power Pool (NWPP)" US department of energy April 2010.
- 70.M. Ehsani, Y. Gao, A. Emadi, Modern Electric, Hybrid Electric, and Fuel Cell Vehicles: Fundamentals, Theory, and Design, Second Edition, CRC Press, 2009.
- 71.M. Choucair, P. Thordarson, and J. A. Stride "Gram-Scale Production of Graphene Based on Solvothermal Synthesis and Sonication", Nature Nanotechnology, 2009, 4, 30.
- 72.M. D. Stoller, S. Park, Y. Zhu, J. An, and R. S. Ruoff "Graphene-Based Ultracapacitors", Nano Letters, 2008, 8, 3498.
- 73.Mingchao Wang, Cheng Yan and Lin Ma "Graphene nanocomposites" http://dx.doi.org/10.5772/50840.
- 74.M.-F. Yu, O. Lourie, M.J. Dyer, K. Moloni, T.F. Kelly, R.S. Ruoff, Strength and Breaking Mechanism of Multiwalled Carbon Nanotubes Under Tensile Load, Science. 287 (2000) 637–640. doi:10.1126/science.287.5453.637.
- 75.M. Shiomi, T. Funato, K. Nakamura, K. Takahashi, M. Tsubota, "Effects of carbon in negative plates on cycle-life performance of valve-regulated lead/acid batteries", J. Power Sources 64 (1997) 147-152 (doi: 10.1016/S0378-7753(96)02515-3).
- 76.Poole, C.P., Jr. and Owens, F.J., "Introduction to Nanotechnology", DOI: 10.1134/S1070427206070366.
- 77.P.P. Prosini, A. Pozio, S. Botti, R. Ciardi, "Electrochemical studies of hydrogen evolution, storage and oxidation on carbon nanotube electrodes", J. Power Sources 118 (2003) 265-269 (doi: 10.1016/S0378-7753(03)00097-1).
- 78.P.T. Moseley, "Consequences of including carbon in the negative plates of Valveregulated Lead–Acid batteries exposed to high-rate partial-state-of-charge operation", J. Power Sources 191 (2009) 134–138 (doi.org/10.1016/j.jpowsour.2008.08.084).
- 79.Phaedon Avouris* "Graphene: Electronic and Photonic Properties and Devices" nano letter DOI: 10.1021/nl102824h |Nano Lett.2010,10, 4285–42.
- 80.Paul Denholm, Jennie Jorgenson, Marissa Hummon, Thomas Jenkin, and David Palchak "The Value of Energy Storage for Grid Applications" National Renewable Energy Laboratory, Task No. SA12.0200 May 2013.

- 81.P. W. Sutter, J-I. Flege, and E. A. Sutter "Epitaxial Graphene on Ruthenium", Nature Materials, 2008, 7, 406.
- 82.Q.-L. Chen, K.-H. Xue, W. Shen, F.-F. Tao, S.-Y. Yin, W. Xu, Fabrication and electrochemical properties of carbon nanotube array electrode for supercapacitors, Electrochimica Acta. 49 (2004) 4157–4161. doi:10.1016/j.electacta.2004.04.010.
- 83.Rachel Carnegie, Douglas Gotham, David Nderitu, and Paul V. Preckel "Utility Scale Energy Storage Systems", State Utility Forecasting Group-June 2013.
- 84.R.H. Baughman, A.A. Zakhidov, W.A. de Heer, Carbon Nanotubes--the Route Toward Applications, Science. 297 (2002) 787–792. doi:10.1126/science.1060928.
- 85.R.J. Brodd, K.R. Bullock, R.A. Leising, R.L. Middaugh, J.R. Miller, E. Takeuchi, Batteries, 1977 to 2002, J. Electrochem. Soc. 151 (2004) K1–K11. doi:10.1149/1.1641042.
- 86.R. M. Westervelt "Graphene Nanoelectronics" Science, New Series, Vol. 320, No. 5874 (Apr. 18, 2008).
- 87.Sanketh R. Gowda, Arava Leela Mohana Reddy, Xiaobo Zhan, and Pulickel M. Ajayan* "Building Energy Storage Device on a Single Nanowire" nano letters dx.doi.org/10.1021/nl2017042 | Nano Lett. 2011, 11, 3329–3333
- 88.S. Mitra, O. Mondal, D. R. Saha, A. Datta, S. Banerjee, and D. Chakravorty "Magnetodielectric Effect in Graphene-PVA Nanocomposites", Journal of Physical Chemistry C, 2011, 115, 14285.
- 89.S. Basu, Ambient temperature rechargeable battery, US4423125 A, 1983. http://www.google.com/patents/US4423125 (accessed May 26, 2015).
- 90.Seoung-Ki Lee, Kuldeep Rana, and Jong-Hyun Ahn* "Graphene Films for Flexible Organic and Energy Storage Devices" dx.doi.org/10.1021/jz400005k| J. Phys. Chem. Lett.2013, 4, 831-84.
- 91.S. Iijima, T. Ichihashi, Single-shell carbon nanotubes of 1-nm diameter, Nature. 363 (1993) 603–605. doi:10.1038/363603a0.
- 92.S. Iijima, Helical microtubules of graphitic carbon, Nature. 354 (1991) 56–58. doi:10.1038/354056a0.
- 93.Sercan Teleke, David Elizondo and Farid Katiraei "Energy Storage Overview: Applications, Technologies, Economical Evolution" Quanta technology.
- 94.S.H. Park, A. Roy, S. Beaupré, S. Cho, N. Coates, J.S. Moon, et al., Bulkheterojunction solar cells with internal quantum efficiency approaching 100%, Nat Photon. 3(2009) 297–302. doi:10.1038/nphoton.2009.69.
- 95.Simon P., Burke A., "Nanostructured carbons: double-layer capacitance and more", The Electrochemical Society Interface, Spring, 2008, 38-43.
- 96.Simon P., Gogotsi Y., "Capacitive Energy Storage in Nanostructured Carbon-Electrolyte Systems", Accounts of Chemical Research 46, 2013, 1094-1098 (doi: 10.1021/ar200306b).
- 97.S. Stankovich, D.A. Dikin, R.D. Piner, K.A. Kohlhaas, A. Kleinhamme, Y. Jia, Y. Wu, S.T. Nguyen, R.S. Ruoff "Synthesis of graphene-based nanosheets via chemical reduction of exfoliated graphite oxide" Carbon 45 [1558–1565] (2007).
- 98.S. Giddey, S.P.S. Badwal, A. Kulkarni, C. Munnings, A comprehensive review of direct carbon fuel cell technology, Progress in Energy and Combustion Science. 38 (2012) 360–399. doi:10.1016/j.pecs.2012.01.003.

- 99.Sungjin Park, Jinho An, Jeffrey R. Potts, Aruna Velamakanni, Shanthi Murali, Rodney S. Ruoff "Hydrazine-reduction of graphite- and graphene oxide" CARBON49 (2011) 3019–3023.
- 100. T. Chen, B. Zeng, J.L. Liu, J.H. Dong, X.Q. Liu, Z. Wu, X.Z. Yang, Z.M. Li "High throughput exfoliation of graphene oxide from expanded graphite with assistance of strong oxidant in modified Hummers method" Journal of Physics : Conference Series 188 (2009).
- 101. T. Ramanathan, A. A. Abdala, S. Stankovich, D. A. Dikin, M. H-Alonso, R. D. Piner, D. H. Adamson, H. C. Schniepp, X. Chen, R. S. Ruoff, S. T. Nguyen, I. A. Aksay, R. K.Prud'Homme, and L. C. Brinson "Functionalized Graphene Sheets for Polymer Nanocomposites", Nature Nanotechnology, 2008, 3, 327.
- 102. T. Zheng, J.N. Reimers, J.R. Dahn, Effect of turbostratic disorder in graphitic carbon hosts on the intercalation of lithium, Phys. Rev. B. 51 (1995) 734–741. doi:10.1103/PhysRevB.51.734.
- 103. T. Zheng, Y. Liu, E.W. Fuller, S. Tseng, U. von Sacken, J.R. Dahn, Lithium Insertion in High Capacity Carbonaceous Materials, J. Electrochem. Soc. 142 (1995) 2581–2590. doi:10.1149/1.2050057.
- 104. Wen Lu* and Rachel Hartman "Nanocomposite Electrodes for High-Performance Supercapacitors", dx.doi.org/10.1021/jz200104n, J. Phys. Chem. Lett.2011, 2, 655–660
- W. Lu, L. Qu, K. Henry, L. Dai, High performance electrochemical capacitors from aligned carbon nanotube electrodes and ionic liquid electrolytes, Journal of Power Sources. 189 (2009) 1270–1277. doi:10.1016/j.jpowsour.2009.01.009.
- 106. W. L. Song, M-S. Cao, M-M. Lu, J. Yang, H-F. Ju, Z-L. Hou, J. Liu, J. Yuan, and L-Z. Fan "Alignment of Graphene Sheets in Wax Composites for Electromagnetic Interference Shielding Improvement", Nanotechnology, 2013, 24, 115708.
- 107. W. S. Hummers Jr and R. E. Offeman "Preparation of Graphite Oxide", Journal of American Chemical Society, 1958, 80 (6), 1339.
- 108. Wonbong Choi*, Indranil Lahiri,1 Raghunandan Seelaboyina, and Yong Soo Kang "Synthesis of Graphene and Its Applications: A Review", Critical Reviews in Solid State and Materials Sciences, 35:52–71, 2010, DOI: 10.1080/10408430903505036.
- 109. Xiaoting Jia,a Jessica Campos-Delgado,b Mauricio Terrones,cd Vincent Meuniere and Mildred S. Dresselhaus* "Graphene edges: a review of their fabrication and characterization", DOI: 10.1039/c0nr00600a, October 2010.
- 110. Xianping Zou, Zongxuan Kang, Dong Shu, Yuqing Liao, Yibin Gong, Chun He, Junnan Hao, Yayun Zhong "Effects of carbon additives on the performance of negative electrode of lead-carbon battery", Electrochimica Acta, Volume 151, 1 January 2015, Pages 89-98 (doi:10.1016/j.electacta.2014.11.027).
- 111. X.Bai, Y. Zhai, and Y. Zhang "Green Approach to Prepare Graphene-Based Composites with High Microwave Absorption Capacity", Journal of Physical Chemistry C, 2011, 115, 11673.
- 112. X. Zhao, Q. Zhang, and D. Chen "Enhanced Mechanical Properties of Graphene-Based Poly (vinyl alcohol) Composites", Macromolecules, 2010, 43, 2357.

- Y. Ando, X. Zhao, H. Shimoyama, G. Sakai, K. Kaneto, Physical properties of multiwalled carbon nanotubes, International Journal of Inorganic Materials. 1 (1999) 77–82. doi:10.1016/S1463-0176(99)00012-5.
- 114. Yun Lu, Sujun Guan, Liang Hao and Hiroyuki Yoshida "Review on the Photocatalyst Coatings of TiO2: Fabrication by Mechanical Coating Technique and Its Application" Coatings 2015, Volume 5, Pages 425-464; (doi:10.3390/coatings5030425)
- 115. Y. Hernandez, M. Lotya, V. Nicolosi, F.M.Blighe, S. De, G. Duesberg, J.N.Coleman "Liquid phase production of graphene by exfoliation of graphite in surfactant/water solutions" Journal of American Chemical Society 131 [3611-3620] (2009).
- 116. Y.-K. Kim, D.-H. Min, Durable Large-Area Thin Films of Graphene/Carbon Nanotube Double Layers as a Transparent Electrode, Langmuir. 25 (2009) 11302– 11306.doi:10.1021/la9029744.
- 117. Y. Sun, Q. Wu, G. Shi, Graphene based new energy materials, Energy Environ. Sci. 4(2011) 1113–1132. doi:10.1039/C0EE00683A.
- 118. Yuyan Shao, Jun Wang, Hong Wu, Jun Liu, Ilhan A. Aksay, Yuehe Lina* "Graphene Based Electrochemical Sensors and Biosensors: A Review" December 2009, DOI: 10.1002/elan.20090057.
- Y. Wu, J. Wang, K. Jiang, S. Fan, Applications of carbon nanotubes in high performance lithium ion batteries, Front. Phys. 9 (2013) 351–369. doi:10.1007/s11467-013-0308-x.
- 120. Y. Zhu, S. Murali, W. Cai, X. Li, J.W. Suk, J.R. Potts, et al., Graphene and Graphene Oxide: Synthesis, Properties, and Applications, Adv. Mater. 22 (2010) 3906–3924. doi:10.1002/adma.201001068.
- 121. Zhihong Luoa,b, Lihua Zhua, Yanfei Huang b, Heqing Tang b,* "Effects of graphene reduction degree on capacitive performances of graphene/PANI composites", Synthetic Metals 175 (2013) 88–96.

List of Publications: Sreedhar Doraswamy - Google Scholar

	Sreedhar Doraswamy Asst. Manager - R&D, Amararaja Batteries; Ph.D Scholar UNISA	Citations h-index i10-index	All 84 5 4	Since 2017 81 5 4
TITLE			CITED BY	YEAR
A CAPSULE T S Doraswamy, S k IN Patent App. 335	YPE ELECTRICAL TEST RIG FOR BATTERY MATERIAL A K, MK S N, J M, VV Rao 9690-001	ND COMPONENT		2021
CONDUCTIVIT S Doraswamy, S k IN Patent App. 339	Y TEST RIG FOR POWDER SAMPLES OF BATTERY 5, MK S N, J M, VV Rao 9688-001			2021
Effect of milled S Doraswamy, K S Materials Today: P	carbon as negative electrode additive for lead acid energy s srinivas, KSN Murthy, M Jagadish, VV Rao Proceedings 38, 3131-3135	torage device	1	2020
Modified Lignos preparation S Doraswamy, S K IN Patent App. 201	sulfonate for Negative Plates of Lead Acid Battery and a Pro K, MK S N, J M, VV Rao 1,941,009,752	cess for its		2019
Synthesis and s D Sreedhar, S Dev Materials Today: P	study of reduced graphene oxide layers under microwave irra vireddy, VR Veeredhi roceedings 5 (2), 3403-3410	adiation	10	2018
Facile synthesis	s and thermal analysis of antimony telluride nanostructures /ireddy, VR Veeredhi		2	2018
Synthesis of de M Nagaraju, K Mol AIP Conference Pr	ndritic-flowers of wurtzite Cu2ZnSnS4 via solvothermal proc kurala, S Mallick, D Sreedhar roceedings 1832 (1), (120030-1) - (120030-3)	ess		2017
Thermal conduct KS Reddy, D Sree International Journ	ctivity of natural fiber, glass fiber & CNTs reinforced epoxy co dhar al of Current Engineering and Technology E-ISSN, 2277-4106	omposites	10	2016
Synthesis of Gr Graphene-NiO VRV Narasimhara Journal of Material	aphene Oxide by Modified Hummers Method and Hydrother Nano Composite for Supercapacitor Application o K, Venkata Ramana G, Sreedhar D Sciences & Engineering 5 (6)	mal Synthesis of	32	2016
Fabrication of C AK Jogi, S D International Journ	Organic Dye Sensitized Solar Cell al of Humanities , Arts, Medicine and Sciences 3 (12), 13-16			2015
Fabrication of N R Mandal, S D, V V International Journ	lano-Hybrid for Constructional Applications v al of Management, Information Technology and Engineering			2015
Supercapacitor SB Gaikwad, S D, International Journ	nanocomposite using rGO, MWNT and PANI V Rao V al of Management, Information Technology and Engineering			2015
Synthesis and e D Sreedhar, KS Re Materials Today: P	evaluation of zinc based nanomaterials for optical-thermal st eddy, VV Rao roceedings 2 (4-5), 3690-3696	udies	1	2015
Synthesis and a D Sreedhar, YV Re Procedia Materials	analysis of ternary oxide nanomaterial for electro-thermal ap eddy, VV Rao : Science 10, 116-123	plications	8	2015

Effect of CNTs on Hardness and Thermogravimetric Properties of LM6 Nanoparticulate Composite KS Reddy, D Sreedhar, GJ raju Materials Today: Proceedings 2 (4-5), 3619-3626	3	2015
Role of reduced graphene oxide on mechanical-thermal properties of aluminum metal matrix nano composites KS Reddy, D Sreedhar, KD Kumar, GP Kumar Materials Today: Proceedings 2, 1270-1275	11	2015
Synthesis and evaluation of zinc based nanomaterials for optical-thermal studies S D, S Reddy K, V Rao V Materials Today: Proceedings 2 (4-5), 3690-3696		2015
Fabrication and Mechanical Properties of AlSi12 Nano Particulate Composite S Reddy K, S D, N P, J Raju G Procedia Materials Science 10, 149-158	4	2015

<u>Annexure - A</u> 1. Comparison of Materials for Lead Acid Battery application

Why Carbon?

S. No	Properties	Carbon	Silver	Copper	Gold	Tungsten	Nickel	Platinum	Tin	Titanium
1	Structure	Hexagonal APF: 0.74; Porous/Non Porous	Face	Face Centered Cubic ; APF: 0.74		BCC; APF: 0.68	FCC; APF: 0.74	FCC; APF: 0.74	BCT; APF: 0.687	Hexagonal APF: 0.74
2	Density (g/cm3)	1.8 - 2.1	10.49	8.96	19.3	19.3	8.9	21.45	7.2	4.5
3	Surface Area (m²/g)	10-1400	1.8	0.02	4	0.18	0.7	22	12	32
4	Conductivity (10 ⁶ S/m)	10-100	63	59.6	41	17.9	14.3	9.43	9.17	2.38
5	Inertness (H ₂ SO ₄)	Unreactive	Reactive Un		reactive	Reactive	Unreactive	Rea	ctive	
6	Melting Point (°C)	3550	961	1084	1064	3422	1455	1768	231	1668

Ref:

- 1. doi: 10.1149/2.062301jes
- 2. doi: 10.1016/j.jpowsour.2017.01.056
- 3. ISBN: 9783642582622
- 4. Patent No: EP0435766A2
- 5. ISBN: 9781118686232

2. Carbon Exploration as Lead acid battery additive Materials



Fig. Classification in terms of usage.

3. Role of Nanocarbon materials in Lead-Acid Battery: Application and Requirements

Nanocarbon (NC) in different formulations play a vital role in different applications. The following table explains various applications and intended benefit from NC.

Applications	Requirements	Nanocarbon Advantage
Advanced SLI (Start Light Ignition)	 90 % - 100% SOC High Discharge current/Low charge current operation 	 Increase in CCA (Cold Cranking Ability) Increase in cold charge current acceptance Improvement in deep discharge recovery due to dark current in modern vehicles Increase in life under deficit charge condition
EFB (Enhanced Flooded Batteries)	 75% - 100% SOC High Discharge current/high Charge current operation 	 Increase in CCA Increase in cold charge current acceptance Improvement in deep discharge recovery due to dark current Increase in SBA life cycle by 30% Increase in 17.5% DOD life cycle by 30% Maintaining higher dynamic charge current throughout the life Increase in charge current acceptance at high SOC – 90%
1) 50% - 90% SOC2) Low toAGM - LABModerate discharge current/ Low charge current operation		 Increase in life Maintaining charging time throughout the life Maintaining dynamic charge acceptance behavior throughout the life Less to assimilate 102 to 105% previous discharge input
Tubular - solar	 20% - 100% SOC Low Discharge current/ Low charge current 	 Increase in life Maintaining charging time throughout the life Maintaining dynamic charge acceptance behavior throughout the life Less time assimilate 102% to 105% previous discharge input

<u>Annexure – B</u>

Lis	t of	Instruments	Used i	in th	is p	proj	ject:	

Equipment	Make	Country	Application
SEM/EDS	Jeol	Japan	Elemental Composition,, Morphological Analysis, Topographical Analysis
XRD Rigaku		Japan	Phase Composition, Crystal 3-D View, Crystalline Structure View, Sample Size Distribution View
ICP-OES	Agilent	Malaysia	Elemental Analysis Capable To Ppb Level
UTM-Universal Testing Machine	Tinius Olsen	U.K	Tensile/Elongation Test, Puncture Resistance Test, Pell Off Test, Flexural Modulus Test
Vickers Hardness Tester	Buhler	Germany	Hardness Measurement, Aging Studies
Stereomicroscope	Seiwa Optical	Japan	Microstructural Analysis, Alloy Characterization
Optical Microscope	Nikon	Japan	Macro And Micro Grain Structural Analysis Of Alloys
LOI	Fire Testing Technologies	U.K	Limiting Oxygen Index For Flame Retardant Poly Materials
	EquipmentSEM/EDSXRDICP-OESUTM-Universal Testing MachineVickers Hardness TesterStereomicroscopeOptical MicroscopeLOI	EquipmentMakeSEM/EDSJeolXRDRigakuICP-OESAgilentVTM-VNiversal resting MachineSubacolVickers Hardness TesterBuhlerStereomicroscopeSeiwa OpticalOptical MicroscopeNikonLOIFirefresting	EquipmentMakeCountrySEM/EDSJeolJapanXRDRigakuJapanICP-OESAgilentMalaysiaJUTM-Universal Testing MachineInius OlsenJuKVickers Hardness TesterBuhlerJapanStereomicroscopeSeiwa OpticalJapanInius OlsenInius OlsenJupanStereomicroscopeSeiwa OpticalJupanInius OlsenSeiwa OpticalSeiwa OpticalInius OlsenSeiwa OpticalJupanInius OlsenSeiwa OpticalSeiwa OpticalInius OlsenSeiwa OpticalSeiwa OpticalInius OlsenSeiwa OpticalSeiwa OpticalInius OlsenSeiwa OpticalSeiwa OpticalInius OlsenSeiwa OpticalSeiwa

9	DSC- Differential Scanning Calorimetry	Netzsch	Germany	Melting And Crystallization, Glass Transition Temperature , Polymer Materials Identification, Degree Of Crystallinity		
10	BET-Surface Area Analyzer	Coulter Corporation	Usa	Surface Area Of Powder Samples, Porosity Measurements		
11	Impedance Analyzer	Wonatech (Zive Sp1)	South Korea	Alloy Characterization, Impedance Measurements.		
12	Electro-Chemical Workstation	Wonatech	South Korea	Potentiostat, Galvanostat, Corrosion Studies, Electro Chemical Behaviors.		
13	Ball Mill	Fritsch	Germany	Size Reduction , Uniform Mixing		
14	Thermo Gravimetric Analyzer	Netzsch	Germany	Thermal Behavior Of Solids, Liquids, Semi Solid Materials		
15	IZOD	Zwick	Germany	Impact Strength Analysis Of Polymers , Alloys, Metals		
16	Wet Laboratory	Alloy Analysis, Separator Analysis, Effluent Water Analysis, Other Battery Components Analysis				
17	Electrical Resistance	Wonatech / Zive Sp1	South Korea	Test Rigs Inhouse Development And Power Source Connected To Wonatech		

$\frac{Annexure - C}{Other outcomes through this project:}$

S.		-			
No.	Title	Status	Remarks		
-	m 1 1 1	0 0 11			
1	To developed	Successfully	Commissioned and have been using for all types		
	the Electrical	designed,	of conductive / Resistive type powder or sheet		
	Test Rigs for	fabricated and	form materials.		
	Powder & Sheet	commissioned			
	Materials	the fixtures.			
2	MWCNT's as	Tests	Paper Communicated to DHET Listed Journal of		
	NAM Additive	successfully	'Current Applied Science & Technology (CAST)'.		
	in Flooded LAB	carried out with			
		12V-65Ah rated			
		flooded battery.			
3	Functionalized	Tested for 12V-	Performance better with gravity cast grid		
	Silica as PAM	3Ah rated	technology were encouraging and tests are under		
	additive in LAB	Battery	progress with stamped grid technology.		
4	Carbon as	Prototype	Successfully replaced the negative lead grid with		
	current	developed with	carbon grid and observed slightly better results		
	collector grid	2V-3Ah cell by	when compared to the control cells. Currently,		
	for LAB	replacing with	under progress of replacing the positive lead grid		
		developed	with carbon grid.		
		carbon grid.			
5	Electrical	Prototype	Tests were carried out with assembling by 3		
	Performance of	developed with	iterations with the coated negative, positive and		
	Graphene	2V-3Ah rated	both the electrodes. It was observed that with the		

	Coated	batteries by	coating of negative electrodes leveraged in higher		
	Electrodes	coating	initial discharge capacity with improved HRD.		
		electrode			
		surface with			
		developed			
		colloidal			
		Graphene			
		solution			
6	Electrical	Prototype	Tests were carried out with assembling by 3		
	Performance of	developed with	iterations with the coated AGM facing the		
	Graphene	2V-3Ah rated	negative, positive and both the electrodes. From		
	Coated AGM	cells by coating	the electrical tests, no significant improvement in		
	Separator	AGM separator	charge acceptance and cold cranking ability. The		
		with developed	PSoC studies have improved with ~20% with both		
		colloidal	side coated separator facing the surface of positive		
		graphene	and negative electrodes.		
		solution			
7	Electrical	Tests were	Carbon coated paper was less effective when the		
	Performance of	successfully	coted side is faced towards the electrode. But		
	Carbon paper	carried out with	shown better adhesion when the coated side was		
	as plate	12V-38Ah rated	faced outward to the electrodes. Later, electrical		
	backing paper	flooded battery.	tests were carried out on the batteries which		
			showed no significant improvement in the quick		
			charge and discharge ability. But, found higher		
			life in light load (~10%) and High Load (~20%).		
8	Nanocarbon	Tests were	Generally, carbon materials are explored in NAM		
	Particles as	successfully	only. But here we attempted to add nanocarbon		

PAM additive in	carried out with	materials in liquid form which are available in the
LAB	12V-38Ah rated	market. Electrical tests were carried out on these
	flooded battery.	batteries which showed no significant
		improvement in the Charge acceptance and cold
		cranking ability. But, found higher life
		improvement in light load (~2 times) which is a
		significant indication to develop for high load
		studies for Hybrid/EV applications.

1. Electrical Analysis of Graphene Coated Electrodes

Graphene Coated Electrode

Fig. Electrodes with and without Graphene Coat.

S. No	Test	Requirem ent	Regular	-Ve Coat	+Ve Coat	+Ve & -Ve Coat
1	C10	95% Output	3.33 (Ah); 111.16%	3.54 (Ah); 118.0%	3.24 (Ah); 108.0%	3.44 (Ah); 114.66%
2	HRD	@ 5Sec (10V)	10.48V	10.23V	9.89V	10.06V
		108s (6V)	106s	160s	132s	125s
3	CAT	0.3A	0.32A	0.36A	0.32A	0.33A

Table. Electrical performance of Electrodes (with and without Graphene Coat).

Inference:

Successfully coated the plates with graphene and tested the batteries (12V-3Ah). Found that the Negative coated plates performed better when compared to the rest in terms of HRD time to reach 6V. And marginal improvement (~4%) in capacity.

2. Electrical Analysis of Graphene Coated Separator



Fig. Process of coating the separator.



PR- Positive Coat wrapping; NR- Negative Coat wrapping;**2R-** Both (+Ve &-Ve) Coat wrapping

Inference:

Successfully coated the separator with graphene and tested the cells (2V) by assembling. Cell with both sides' separator found to be better with one side

3. Analysis of Carbon as current collector grid (Preliminary Study: Replaced negative Lead Grid)



S. No.	Parameters	Regular (-Ve)	Trail
1	Base Element	Lead (Pb)	Carbon ©
2	Grid Weight (gm)	10	0.5

Fig. Pure Carbon sheet.



Regular Lead Grid





Fig. Capacity (C10) Discharge/Charging Profile of regular and caron grid cell.

Inference:

Successfully replaced negative lead grid with carbon grid and found to be superior when compared to lead grid in C10capacity test.

Test	Requirement	Regular	Carbon Grid	Remarks
Capacity (C10)	≥95%	3.08(Ah); 103%	3.3 (Ah); 110%	~7% Improvement